EVOLUTION OF PHOTOPERIODISM IN THE THREESPINE STICKLEBACK,

GASTEROSTEUS ACULEATUS

by

CONOR SAVAGE O'BRIEN

A DISSERTATION

Presented to the Department of Biology and the Graduate School of the University of Oregon in partial fulfillment of the requirements for the degree of Doctor of Philosophy

December 2011

DISSERTATION APPROVAL PAGE

Student: Conor Savage O'Brien

Title: Evolution of Photoperiodism in the Threespine Stickleback Gasterosteus aculeatus

This dissertation has been accepted and approved in partial fulfillment of the requirements for the Doctor of Philosophy degree in the Department of Biology by:

Eric Johnson	Chairperson
William Cresko	Advisor
William Bradshaw	Member
Judith Eisen	Member
Patricia McDowell	Outside Member
and	
Kimberly Andrews Espy	Vice President for Research & Innovation/Dean of the Graduate School
Original approval signatures	are on file with the University of Oregon Graduate School.

Degree awarded December 2011

© 2011 Conor Savage O'Brien

DISSERTATION ABSTRACT

Conor Savage O'Brien Doctor of Philosophy Department of Biology December 2011 Title: Evolution of Photoperiodism in the Threespine Stickleback *Gasterosteus aculeatus*

In seasonal environments, the ability to take advantage of the favorable seasons and avoid or mitigate the effects of the unfavorable ones is essential for organismal fitness. Many polar and temperate organisms use photoperiod (length of day) to time seasonal life history events because photoperiod's regular annual cycle makes it a very reliable indicator of seasonality. This reliability allows organisms to anticipate and properly prepare for seasonal change. Although photoperiodism is widespread in polar and temperate vertebrates, little is known relative to invertebrates regarding how its use varies with environment and this method's underlying genetic and physiological basis. This dissertation is focused on demonstrating the proper methodology for the study of photoperiodism and establishing the threespine stickleback as a model of vertebrate photoperiodism.

Chapter I is an introduction to photoperiodism, how it is influenced by environment, the physiological basis of its output, and a summary of the chapters that follow. Chapter II explains an analytical method to test for causality and applies this method to data that have been interpreted as evidence that the circadian clock is causally involved in photoperiodism. Chapter III describes the photoperiodic response of threespine stickleback *Gasterosteus aculeatus* populations from two latitudes. These

iv

results are used to inform an empirical examination of the previously described assertion that the circadian clock is causally involved in photoperiodism. Chapter IV examines the physiological basis of early photoperiodic response using the threespine stickleback as a model teleost fish. Chapter V summarizes the previous chapters, describes their significance, and suggests future research directions.

This dissertation includes both previously published and co-authored material. Supplementary Excel files demonstrating the analyses used in Chapter III are also included in this dissertation.

CURRICULUM VITAE

NAME OF AUTHOR: Conor Savage O'Brien

GRADUATE AND UNDERGRADUATE SCHOOLS ATTENDED:

University of Oregon, Eugene OR Bowdoin College, Brunswick ME

DEGREES AWARDED:

Doctor of Philosophy in Biology, 2011, University of Oregon Bachelor of Science in Biology, 2003, Bowdoin College

AREAS OF SPECIAL INTEREST:

Evolutionary genetics Chronobiology

PROFESSIONAL EXPERIENCE:

Research assistant, Boston University School of Medicine, 2003 - 2005

GRANTS, AWARDS, AND HONORS:

NSF IGERT Fellow, 2005 – 2007

NIH Genetic Training Grant Fellow, 2007 – 2009

PUBLICATIONS:

O'Brien, C. S., Bradshaw, W. E., & Holzapfel, C. M. (2011). Testing for causality in covarying traits: genes and latitude in a molecular world. *Molecular Ecology* **20**, 2471--2476.

Yeates-Burghart, Q. S., **O'Brien, C. S.**, Cresko, W. A., Holzapfel, C. M., & Bradshaw, W. E. (2009). Latitudinal variation in photoperiodic response of the three-spined stickleback *Gasterosteus aculeatus* in western North America. *Journal of Fish Biology* **75**, 2075-2081.

Lou D., Dominguez I., Toselli P., Landesman-Bollag E., **O'Brien C. S.**, Seldin D. (2008). The alpha catalytic subunit of protein kinase CK2 is required for mouse embryonic development. *Molecular and Cellular Biology* **28**, 131-139.

ACKNOWLEDGMENTS

This dissertation was only possible with the help of friends, family and colleagues. Their contributions to the work presented herein, my intellectual growth as a graduate student and to my time in Oregon speaks to the general excellence of those around me.

My understanding of biology, especially the fields of chronobiology and evolutionary genomics, has been shaped by countless discussions with fellow lab mates, members of the Institute of Ecology and Evolution and the NSF IGERT Evolution of Development Journal Club. My dissertation committee members, B. Bradshaw, W. Cresko, C. Holzapfel, J. Eisen, E. Johnson, and P. McDowell, has always been supportive and insightful, even as I confused them and myself with quantitative PCR results.

In many cases, the data presented herein is the result of collaboration. I would especially like to Quick Yeates, for the countless hours we spent together and her support while dealing with the frustrations associated with establishing a model. C. Zimmerman deserves mention for contributions toward our understanding of threespine stickleback life history in Oregon. Many undergraduates made contributions, but Larissa Unruh and Ryan Bourdo deserve specific mention. I enjoyed mentoring them through the frustrations and successes of research. The data they generated herein is testament to their perseverance. Finally, I would like to acknowledge the logistical support of the Oregon Institute of Marine Biology, without which the data presented in Chapters III and IV would have been much more difficult to gather.

It has been a pleasure to be advised by William E. Bradshaw, William A. Cresko and Christina M. Holzapfel. Their approaches are distinct, but they share a dedication to

viii

quality work and an unwavering enthusiasm for science. I hope to reflect these characteristics in future endeavors.

The work presented herein was possible through generous funding from the National Science Foundation IGERT training grant DEG_0504727 and the National Institutes of Health training grant 5-T32-GMO7413. Additional funding support came from NSF grants IOS_0818738, IOS_1027283 and DEB_0949053 to William A. Cresko.

This dissertation is dedicated to my grandmothers. To my maternal grandmother, Elizabeth Savage Hart, the first of my family to study biology in Oregon and a true *tierfreundin*: living here has helped me discover more of you within me. I now understand why you always smiled when the conversation turned to Oregon. And to my paternal grandmother, Ruth Hazel O'Brien: for the grace she brings to our family. Your intelligence and resolve are wonderful to watch and, hopefully, internalize.

Chapter	Page
I. INTRODUCTION	1
The Importance of Photoperiodism	1
What Is the Relationship between the Circadian Clock and Vertebrate Photoperiodic Time Measurement?	3
How Does Geography Influence Vertebrate Photoperiodism?	8
How Conserved Is the Physiological Basis of Vertebrate Photoperiodic Response?	10
Mammals	14
Birds	15
Fishes	18
Section Summary	23
Brief Outline of this Dissertation	24
II. TESTING FOR CAUSALITY IN COVARYING TRAITS: GENES AND LATITUDE IN A MOLECULAR WORLD	29
Bridge	39
III. GEOGRAPHY OF THE CIRCADIAN GENE <i>CLOCK</i> AND PHOTOPERIODIC RESPONSE IN WESTERN NORTH AMERICAN POPULATIONS OF THE THREESPINE STICKLEBACK	40
Introduction	40
Materials and Methods	43
Photoperiodic Response	43
Clock Polyglutamine Domain	44
Analyses	46

Chapter

Page

Results	47
Photoperiodic Response Is Very Similar Across Populations at Northern and Southern Latitudes	47
The Polyglutamine Domain of <i>clock</i> Varies Across Individual Stickleback but Shows No Population Structuring	48
Discussion	49
Stickleback Have Similar Photoperiodic Responses at Northern and Southern Latitudes	49
Absence of <i>clock</i> Polyglutamine Domain Length (polyQ)	51
Bridge	55
IV. CONSERVATION OF THE PHOTOPERIODIC NEUROENDOCRINE AXIS AMONG VERTEBRATES: EVIDENCE FROM THE TELEOST FISH, <i>GASTEROSTEUS ACULEATUS</i>	57
Introduction	57
Materials and Methods	62
Gasterosteus aculeatus Stocks	62
Experimental Design	63
Target Gene Identification	64
mRNA <i>in-situ</i> hybridization	66
Quantitative Real-Time PCR (qPCR)	67
Results	67
The Genomic Location and Annotation of Hypothesized Targets of Photoperiodism	68
TSHβ1 Expression Is Localized to Expected Regions of the Brain	70

Chapter

Quantitative Real-Time PCR (qPCR) Shows Rapid Response of the Target Genes to Stimulatory Photoperiods	. 71
Discussion	. 74
Answers to Primary Questions	. 74
Thyroid Stimulating Hormone	. 74
Gonadotropin Releasing Hormone	. 75
Luteinizing Hormone	. 77
Conclusions	. 77
Bridge	. 77
V. CONCLUSION	. 80
How Should a Complex Trait like Vertebrate Photoperiodism Be Studied?	. 80
Is the Hormonal Basis of Photoperiodic Response Conserved Among Vertebrates?	. 83
What Makes a Good Model of Vertebrate Photoperiodism?	. 84
How Will the Threespine Stickleback Inform Our Understanding of Photoperiodism in Relation to Life History and Local Environment?	. 87
How Will the Threespine Stickleback Inform Our Understanding of the Genetic Basis of Photoperiodism?	. 88
Conclusion: How Studies of Vertebrate Photoperiodism Will Inform Our Understanding of Evolution	. 89
APPENDICES	. 92
A. SUPPLEMENTARY INFORMATION: CHAPTER II	. 92
B. SUPPLEMENTARY INFORMATION: CHAPTER III	. 93
C. SUPPLEMENTARY INFORMATION: CHAPTER IV	. 94

Page

Chapter

REFERENCES CITED	96
------------------	----

SUPPLEMENTARY FILES

EXCEL FILE: S.1 CALCULATION OF RESIDUALS EXCEL FILE: S.2 CALCULATION OF RESIDUALS FOR POLYQ LENGTH

LIST OF FIGURES

Figure	Page
1.1. The photoperiodic neuroendocrine axis	12
2.1. Inference of causality between circadian rhythmicity and photoperiodism	
due to their common covariation with the independent variable latitude	30
2.2. Use of residuals to test for a causal relationship between	
circadian rhythmicity and photoperiodism in Drosophila littoralis	33
2.3. Verification of analysis of residuals as a test for a causal	
relationship between photoperiodism and circadian rhythmicity	34
2.4. Latitudinal covariation of mean OtsClk1b Poly Q domain length (Poly Q) and	
run (migration) time in Chinook salmon.	35
3.1. Photoperiodic response of male and female threespine stickleback	48
3.2. Polyglutamine domain (PolyQ) in the <i>clock</i> gene in southern and northern	
populations of <i>G. aculeatus</i>	49
3.3. Day length and temperature profiles during the winter and spring in	
Oregon and Alaska	51
4.1. The photoperiodic TH pathway in fish, mammals, and birds	57
4.2. The threespine stickleback genome contains two TSHβ paralogs	67
4.3. The threespine stickleback genome contains two GnRH orthologs	68
4.4. The threespine stickleback genome contains a single subunit $LH\beta$	69
4.5. TSHβ1 expression in the pars distalis of the pituitary	70
4.6. The effect of photoperiod on the thyroid hormone pathway	72

LIST OF TABLES

Table	
4.1. A two-way ANOVA for TSH β 1, GnRH3, and LH β expression	72

CHAPTER I

INTRODUCTION

The importance of photoperiodism

Our planet experiences two great environmental rhythms. The daily cycle of light and dark is caused by the rotation of the earth about its axis, and the seasonal cycle in climate is caused by the angle of the earth's axis relative to its plane of orbit. Organisms must be adapted to the effects of these rhythms to maximize evolutionary fitness.

Some prokaryotes and all eukaryotes have circadian rhythms (from the Latin circa, meaning "around" and diem meaning "day") that control daily organismal processes. Circadian rhythms are endogenous (internal, self-sustained) and set by the daily light/dark cycle (Dunlap *et al.*, 2004). Organismal behavior and development that occurs on a seasonal scale is often mediated by photoperiod, or length of day. Photoperiodism is widespread among temperate and polar organisms (Goldman *et al.*, 2004; Bradshaw & Holzapfel, 2007). Empirical results demonstrate that proper functioning of these timekeeping mechanisms is critical to organismal fitness in the context of daily (Emerson *et al.*, 2008; Yerushalmi *et al.*, 2011) and seasonal (Bradshaw *et al.*, 2004) rhythms.

Fitness in seasonal environments depends on an organism's ability to exploit the favorable season (e.g. to maximize growth and reproduction), avoid or mitigate the

effects of the unfavorable season (e.g. through dormancy and migration) and to make a timely transition between the two life history modes (Bradshaw *et al.*, 2004; Bradshaw & Holzapfel, 2007). Photoperiod is widely used for this purpose because it has a highly reliable annual cycle. At any given latitude, its cycle is exactly the same from year to year, unlike other environmental indicators of seasonality, such as temperature or rainfall. The reliability of the photoperiodic cycle means that it can be used as an anticipatory cue of seasonal change. Transitions between life history modes take time, and the ability to anticipate future conditions means that these transitions can be completed at the optimal time of year.

In shorter-lived organisms, such as invertebrates, absolute photoperiod tends to govern a go/no-go response that once initiated, runs to completion (reviewed in Bradshaw & Holzapfel, 2007). In longer-lived organisms, such as most vertebrates, photoperiodic response results from an interaction between photoperiod and a circannual rhythm (Bradshaw & Holzapfel, 2007). These are physiological rhythms endogenous to the organism that persist under constant temperature and photoperiod (Goldman *et al.*, 2004). They are set by the natural photoperiod cycle and produce an annual rhythm in photosensitivity and photorefractoriness. Long or increasing photoperiods initiate the go/no-go response during photosensitive phases. The refractory phases cause organisms to be non-responsive to photoperiods that would otherwise be stimulatory, thus allowing time to properly prepare for seasonal change (Bradshaw & Holzapfel, 2007). Short photoperiods reset the circannual rhythm (Goldman *et al.*, 2004).

Photoperiodism is widespread in Gnathostomata (the "higher" vertebrates), but its distribution in more basal vertebrate lineages is largely unknown. There is a lack of

studies in Cephalochordates, Agnathans, Chondrichthyes, as well as the basal Actinopterygii lineages. The few studies that have been conducted focused on a single life history event and none examined the effect of photoperiod on sexual maturation or reproduction (reviewed by Bradshaw & Holzapfel, 2007). At least one basal sarcopterygian, the Australian lungfish, is photoperiodic (Kemp, 1984), but this appears to be the sole study of photoperiodism in basal Sarcopterygii.

Photoperiod provides the go/no-go signal for sexual maturation in many species of teleost fishes, mammals and birds (Bradshaw & Holzapfel, 2007), although it also affects other phenotypes, including the physiological processes associated with migration and dormancy (Bromage *et al.*, 2001; Dawson *et al.*, 2002; Goldman *et al.*, 2004). Despite the prevalence of photoperiodism within vertebrates and its effects on important phenotypes, fundamental questions remain unresolved. This dissertation is focused on developing a vertebrate model that can address these questions, which range from the involvement of photoperiodism in life history, to the physiological and genetic processes that underlie its phenotypic outputs.

What is the relationship between the circadian clock and vertebrate photoperiodic time measurement?

The circadian clock is the endogenous time keeping mechanism that is the basis of daily rhythms in organismal processes ranging from gene expression to behavior. It is entrained by the daily light/dark cycle. Photoperiodic time measurement controls seasonal processes and is entrained by either the absolute photoperiod (day length) or change in photoperiod. It is intuitive to suggest that the circadian clock forms the

mechanistic basis of photoperiodic time measurement as both are entrained by properties of the daily light cycle. If true, a single mechanism would be responsible for the timing of behavior and development on both daily and seasonal scales.

This hypothesized connection between the circadian clock and photoperiodic time measurement has been long-standing (Bünning, 1936) and over time has become assumed in the literature (e.g. Hazlerigg & Loudon, 2008). The two time keeping mechanisms have a causal relationship in plants (Wilczek *et al.*, 2009; Kobayashi & Weigel, 2007) and in a laboratory strain of Syrian hamsters (Shimomura *et al.*, 1997; Lowrey *et al.*, 2000), but it remains unknown if they are causally connected in animal populations that exist in nature.

If the circadian clock is the basis of photoperiodic time measurement, we expect a correlation between photoperiodic response and genetic variation in aspects of the circadian clock, such as period or amplitude of the circadian rhythm, or duration or timing of circadian activity. To our knowledge, this potential correlation has only been studied in outbred populations of one vertebrate. Northern populations of the white-footed mouse, *Peromyscus leucopus*, may exhibit gonadal regression in response to short day lengths, while those from southern populations are unaffected and breed year round (Lynch *et al.*, 1981; Heideman *et al.*, 1999). The period of circadian rhythm does not differ under constant darkness, long days or short days among these populations (Carlson *et al.*, 1989). Selection for increased and decreased photoperiodic response in lines derived from a single population produced similar results (Majoy & Heideman, 2000). Thus, neither genetic variation in photoperiodism within or among populations of white-footed mice is correlated with circadian rhythmicity, nor does selection on photoperiodic

response produce a corresponding change in circadian rhythmicity. Although we must be cautious in extrapolating from studies of a single vertebrate, these results support independence of the circadian clock and photoperiodic time measurement.

Correlation between allelic variation in the *clock* gene, and variation in the timing of seasonal traits mediated by photoperiodic response, has been interpreted as support for a circadian clock – photoperiodic time measurement connection. The Clock protein is a basic-helix-loop-helix transcription factor with four protein domains: a DNA binding domain, two protein dimerization domains and a transactivation domain that is characterized by a carboxyl-terminal repeat motif (Darlington *et al.*, 1998). The daily circadian rhythm is set by the expression of *clock*, which autoregulates its own transcription through a negative feedback loop (Darlington *et al.*, 1998). This series of molecular interactions is well characterized in mammals. It appears to have the same function in other vertebrate taxa, including teleost fishes and birds, but this has not been directly tested (Helfer *et al.*, 2006; Vatine *et al.*, 2011). Differences in the number of glutamine repeats in the transactivation domain may affect the circadian rhythm by altering the ability of *clock* to promote transcription (Darlington *et al.*, 1998; Gekakis *et al.*, 1998; Avivi *et al.*, 2001).

Length variation of the *clock* polyglutamine transactivation domain (hereafter termed polyQ) has been proposed as the mechanism by which the circadian clock, via its hypothesized role in photoperiodic time measurement, affects seasonal timing in several species (O'Malley *et al.*, 2010). An association between polyQ domain length of *OtsClock1b*, a *clock* paralog specific to the salmonid lineage and run timing (the seasonal timing of upstream migration in freshwater) was found in two of four Pacific salmon

species examined (O'Malley *et al.*, 2010): the stronger the latitudinal cline in run time, the stronger the latitudinal cline in polyQ domain length. Comparison of *Otsclock1b* allele frequency and neutral genomic loci suggested that these clines were maintained by divergent selection (O'Malley & Banks, 2008a; O'Malley *et al.*, 2010). As migration run time in anadromous salmon is mediated through photoperiodic entrainment of the endogenous circannual rhythm (Quinn & Adams 1996; Bromage *et al.*, 2001), these data were interpreted as evidence that *OtsClock1b* mediates seasonal adaptation by affecting photoperiodic time measurements (O'Malley *et al.*, 2010).

A similar association was found in European populations of a non-migratory passerine. *Clock* polyQ length increases with latitude in the blue tit *Cyanistes caeruleus* (Johnsen *et al.*, 2007). As with the Pacific salmonids, this cline also appears to be maintained by selection (Johnsen *et al.*, 2007). A follow up study in the same populations demonstrated an association between polyQ domain length and breeding time; female blue tits with shorter alleles tended to breed earlier in the season (Liedvogel *et al.*, 2009). Because photoperiodic response controls the timing of blue tit reproductive behavior (Lambrechts *et al.*, 1997), these studies concluded *clock* affects seasonal timing through a causal role in photoperiodic time measurement (Johnsen *et al.*, 2007; Liedvogel *et al.*, 2009).

The proposed connection between allelic variation in a central circadian clock gene and photoperiodism is tantalizing because it supports the hypothesis of a causal connection between the two biological clocks, and suggests a mechanism for variation in photoperiodic response. It is premature, however, to conclude that the circadian clock is involved in photoperiodic time measurement. These studies implicitly assumed that any

effect of *clock* allelic variation on photoperiodic time measurement would be through its role in the circadian clock. This is an unreasonable assumption, as we do not know the role of the *clock* gene in the teleost circadian clock and the possibility of a pleiotropic function for the *clock* gene, independent of its role in the circadian clock, is not considered. In addition, these studies were limited by their reliance on phenological variation as a proxy for variation in photoperiodic response. Run timing in chum (*Oncorhynchus keta*) and Chinook salmon (*Oncorhynchus tshawytscha*), the two species showing the strongest associations between latitude and *OtsClock1b* polyQ, is mediated by photoperiod (Clarke *et al.*, 1994; Quinn & Adams, 1996), but proximate environmental factors, such as water temperature and flow rate can affect run timing, especially in Chinook (Crozier *et al.*, 2008). Likewise, the timing of egg laying in the blue tit is largely controlled by photoperiodic response (Lambrechts *et al.*, 1997), but it is also influenced by local climate and food availability (Westwood & Murton, 1997).

Chapter II describes a simple, but underutilized method that can be used a priori to determine if further investigation of *clock* allelic variation and run timing is appropriate. This genetic variation and phenotypic variation are assumed to have a causal (i.e. necessary) relationship with one another because of their covariation across a latitudinal gradient. If *clock* allelic variation and run timing have a causal relationship, the correlation between them should remain if the common effect of latitude is removed. If the causal relationship does not remain after this test, further consideration of the relationship is inappropriate and tests of the assumptions underlying the hypothesized connection between the circadian clock and photoperiodic time measurement should not be conducted in this context.

Chapter III examines the photoperiodic response of multiple threespine stickleback populations from two latitudes in controlled conditions independent of other environmental signals that may affect response. Allelic variation of the *clock* gene polyQ domain is also assessed in these populations. The results are discussed in the context of methodological approaches to understanding the genetic basis of seasonal timing and the benefits of assessing photoperiodic response directly for such studies.

How does geography influence vertebrate photoperiodism?

As latitude increases, length of the favorable season decreases, resulting in greater consequences for the mistiming of seasonal development or behavior at higher latitudes. Thus, reliance on photoperiod as a seasonal timer is expected to increase with latitude, while the effects of photoperiod on southern populations are more likely to be mediated by other environmental characters, such as temperature.

This trend is well supported by intraspecific comparisons along latitudinal gradients in insects (reviewed by Danilevskii, 1965). For instance, the correlation between latitude and the photoperiod that terminates diapause is \geq .95 in the pitcher plant mosquito *Wyeomyia smithii* (Bradshaw & Holzapfel, 2001). The data available from vertebrates is limited, but also supports this trend. As latitude increases, photoperiod has an increasing effect on metabolic traits in three species of lizards (Lashbrook & Livezey, 1970; Angilletta, 2001; Uller & Olsson, 2003). In northern populations of Scandinavian frogs, photoperiod strictly regulates the timing of reproduction, while in southern populations it interacts with temperature to regulate the timing of reproduction (Laurila *et al.*, 2001). As latitude increases, non-stimulatory photoperiods have increasing control

over gonadal regression in two species of Peromyscus mice (Lynch *et al.*, 1981; Sullivan & Lynch, 1986; Heideman *et al.*, 1999; Lowrey *et al.*, 2000), and embryonic dormancy in several mustelid species (Thom *et al.*, 2004).

Conclusions regarding the effects of latitude on the prevalence of photoperiodism in other vertebrate taxa have been limited for two reasons. First, many studies of vertebrate photoperiodism use model animals that are difficult to raise and maintain in the lab and are impractical to manipulate in experimental conditions that control for the effects of environmental signals beyond photoperiod. Instead, these studies rely upon assumed proxies of photoperiodic time measurement, such as timing of migration, seasonal quiescence, or reproduction. Although these traits are mediated by photoperiodic response, other, unmeasured environmental signals, such as temperature, nutritional availability and presence of con-specifics can affect the timing of their expression (Crozier et al., 2008; Dawson, 2002). Although this type of study is often necessary for non-model organisms that may be of ecological or economic importance, the potential effects of these unmeasured environmental signals can confound interpretation of results. This has limited our ability to understand among population variation in the timing of photoperiodically mediated traits and its effects on life history and local adaptation to the environment.

A second type of study utilizes model organisms that can be studied in laboratory conditions, but only considers, or is only able to consider, individuals from a single, laboratory maintained population. This type of model has produced considerable insights into the transcriptional, hormonal and anatomical responses to photoperiod (Yasuo & Yoshimura, 2009), but the focus on single populations as representative of a species has

limited our ability to understand the genetic and physiological basis of photoperiodic response in natural animal populations, and how it may vary among populations inhabiting different environments.

Bridging the gap between these two types of vertebrate photoperiodic models a model organism with a large natural range that can be raised, maintained, and manipulated in controlled conditions to eliminate the potential effects of environmental signals besides photoperiod. In addition, it must be amenable to the hormonal and genetic techniques necessary to examine the mechanisms that underlie the phenotypic response to photoperiod. Chapter III describes the phenotyping of photoperiodic response in populations of threespine stickleback, a teleost fish found across a wide latitudinal range. These data demonstrate that the threespine stickleback meets these criteria for a model of vertebrate photoperiodic response.

How conserved is the physiological basis of vertebrate photoperiodic response?

The physiological basis of photoperiodic response appears to be conserved across vertebrates. The role of thyroid hormone (TH) in reproduction may have originated in basal chordates (Heyland *et al.*, 2005; Paris *et al.*, 2010) and its control of the photoperiodic initiation of sexual maturation is remarkably similar between mammals and birds (Yasuo & Yoshimura, 2009). Its role in teleost fishes, the most speciose vertebrate clade, is less clear. A full understanding of the role of the TH pathway across vertebrates requires a teleost model in which the physiological basis of photoperiodic response can be studied using hormonal and anatomical techniques. The threespine stickleback is an ideal model for such studies because we have shown that it has a strong

photoperiodic response, which can be assessed in isolation from other environmental factors (Chapter III) and is amenable to techniques necessary to make direct comparisons to mammal and bird models.

Initiation of sexual maturation via photoperiodic response has two basic elements: (1) reception and encoding of the photoperiod signal and (2) a neuroendocrine axis that stimulates gonadotropin production. The anatomical and hormonal basis of these elements will be described in mammals and compared to birds and teleost fishes to highlight areas of conservation, divergence, and those that require study.

Figure 1.1 is a comparative diagram of this process that complements the following sections. It illustrates the high degree of conservation in the hormonal and anatomical basis of the neuroendocrine axis in mammals and birds and the relative lack of comparable data in photoperiodic teleost fishes.



Legend







Fig. 1.1. The photoperiodic neuroendocrine axis.

Mammals

Signal reception and encoding

In mammals, the photoperiodic signal is received by the retina and neuronally communicated through the suprachiasmatic nuclei (SCN), the paraventricular nucleus (PVN), and the superior cervical ganglion (SCG) to the pineal gland (Moore *et al.*, 1995). Melatonin production from the pineal gland is inhibited by light and permitted in darkness, creating a daily rhythm where concentration peaks at night. Thus, melatonin secreted from the pineal gland encodes the photoperiodic signal and communicates it to the neuroendocrine axis that mediates sexual maturation (Cassone *et al.*, 1998).

Neuroendocrine axis

A single stimulatory photoperiod (hereafter referred to as a "long day") is sufficient to invoke hormonal response of the neuroendocrine axis. In the pars tuberalis of the anterior pituitary, a long day melatonin rhythm stimulates expression of the transcription factor Eya3 and its coactivators (Dardente *et al.*, 2010; Masumoto *et al.*, 2010). These induce expression of the thyroid stimulating hormone (TSH) β and α subunits (Hanon *et al.*, 2008; Ono *et al.*, 2008; Yasuo *et al.*, 2010). Like the gonadotropins, TSH is heterodimer consisting of protein-specific β subunit and an α subunit (GTH α) that is common to TSH and all gonadotropins.

This long day response is only induced in pars tuberalis cells expressing the melatonin receptor MT1 (Dardente *et al.*, 2003). In fact, the MT1 receptor may be the sole mechanism by which the neuroendocrine axis receives the photoperiodic signal: photoperiod mediates expression of MT1 in the pars tuberalis (Dardente *et al.*, 2003),

especially as the second melatonin receptor subtype is nonfunctional in at least one photoperiodic mammal (Weaver *et al.*, 1996).

TSH travels to the mediobasal hypothalamus (MBH). Here it stimulates production of Deiodinase 2 (*dio2*) from tanycytes lining the third ventricle (Watanabe *et al.*, 2004; Yasuo *et al.*, 2005; Yasuo *et al.*, 2007). One of three deiodinases found in bony vertebrates, the main role of *dio2* is to deiodinate thyroxine (T4) to triiodothyronine (T3). T4 is produced in the thyroid gland and transported via the hypophyseal portal system to the MBH. Here, Dio2 catalyzes the conversion of T4 to T3, which is several times more biologically active than T4.

T3 acts locally to stimulate production of gonadotropin releasing hormone (GnRH) from neurons in the MBH, probably through direct action on the neuronal dendrites. All mammals have GnRH1 and GnRH2 and some have lamprey GnRH3 (l-GnRH3) (Yahalom *et al.*, 1999; Hiney *et al.*, 2002). GnRH1 is the primary hypophysiotropic paralog in mammals as it is primarily expressed in preoptic neurons that have extensive projections to the median eminence (Dubois *et al.*, 2002). GnRH2 is primarily found in the midbrain, although its distribution can overlap with that of GnRH1 (Dubois *et al.*, 2002). It is thought to function as a neuromodulator and may play a role in regulating sexual behavior (Millar, 2005). The function of l-GnRH3 is less clear. Its distribution largely overlaps with GnRH1 and it may regulate FSH secretion in some mammals (Hiney *et al.*, 2002), although the effect of photoperiod upon it has not been studied.

GnRH is released from axon terminals in the median eminence into the pituitary portal system. The portal system transports the GnRH to the pars distalis of the pituitary

(Yasuo & Yoshimura, 2009), where it stimulates gonadotropin secretion in cells expressing the GnRHI receptor (Millar, 2005). Seasonal plasticity of GnRH1 receptor expression varies amongst photoperiodic mammals (Ciechanowska *et al.*, 2008; Townsend *et al.*, 2009). This expression pattern suggests a role in mediating photoperiodic response, but to our knowledge, this has not been studied.

The gonadotropins, follicle stimulating hormone (FSH) and luteinizing hormone (LH), are heterodimers of protein specific β subunits and GTH α . They are secreted from gonadotropic cells in the pars distalis and released into the bloodstream where they act upon peripheral targets. Their main targets are the gonads, where they stimulate production of the sex hormones. In mammals, the main steroid sex hormones are estradiol in females and testosterone in males (which is intracellularly aromatized at the site of action to estradiol). Feedback of sex hormones on gonadotropin expression is an important mediator of sexual maturation (Shupnik, 1996).

Birds

Signal reception and encoding

In birds, photoperiodic signal reception is extraretinal and does not involve melatonin or the pineal gland (Sharp, 2005). Opsin 5 is expressed in neurons located in the paraventricular organ (PVO) of the photoperiodic Japanese quail (Nakane *et al.*, 2010). These neurons project from the PVO to the median eminence. The median eminence is adjacent to the pars tuberalis of the pituitary, which is the anatomical location of the first hormonal response to a stimulatory photoperiod, which suggests a pathway for photoperiod signal transduction to the neuroendocrine axis. In addition, the peak wavelength absorbance of Opsin 5 is within the range known to elicit gonadotropin release in quail (Nakane *et al.*, 2010), further suggesting it is the receiver of the photoperiod signal.

Although these results strongly support Opsin 5 as a mechanism for light reception, both melanopsin and vertebrate ancient opsin have similar absorbance spectra and distributions within the brain (Chaurasia *et al.*, 2005; Halford *et al.*, 2009). Given current data, the role of these opsins and that of Opsin 5 in photoperiod signal reception cannot be confirmed.

Neuroendocrine axis

Reception and communication of the photoperiodic signal differs between mammals and birds, but the neuroendocrine axis itself is broadly conserved. As in mammals, the earliest response to a long day occurs in the pars tuberalis, where TSH pulses during the first shortened night (Nakao *et al.*, 2008). As in mammals, the TSH pulse coincides with a pulse in the transcription factor *eya3* expression (Nakao *et al.*, 2008), which suggests a conserved role for its inducement of TSH expression. These pulses during the first shortened night are followed shortly by an increase in *dio2* expression around the third ventricle and a corresponding decrease in dio3 (Yasuo *et al.*, 2005; Nakao *et al.*, 2008). Increased TSH binding around the third ventricle during this initial response suggests it is driving the observed increase in *dio2* (Nakao *et al.*, 2008).

Long day responses of the thyroid hormones in birds mirror those in mammals. Long days cause an increase in T4 and T3 levels in the MBH, although their initial long day response has not been studied (Yoshimura *et al.*, 2003). Dio2 in the MBH catalyzes the local conversion of T4 to the bioactive T3 (Yoshimura *et al.*, 2003; Yasuo *et al.*, 2005).

T3 stimulates hypophysiotropic GnRH in long day conditions. Birds contain GnRH1 and GnRH2 and, at least in some species, l-GnRH3 (Dubois *et al.*, 2002; Bentley *et al.*, 2004). GnRH1 and l-GnRH3 neurons are distributed in the preoptic area and have extensive projections to the median eminence (Dubois *et al.*, 2002), consistent with a hypophysiotropic function. GnRH2 is found in the midbrain and is thought to function primarily as a neuromodulator (Dubois *et al.*, 2002). As with mammals, the function of l-GnRH3 is largely unknown.

Under long days, GnRH nerve terminals move closer to the boundary of the median eminence, a process that is induced by T3 (Yamamura *et al.*, 2004; 2006). Presumably, T3 acts locally upon the GnRH neurons after it is catalyzed from T4 in the MBH. This change in terminal positioning could facilitate release of GnRH into the median eminence, where it can be transported to the pituitary. It is unknown if mammalian GnRH release is facilitated by the same mechanism, although this would be expected given the conservation of the photoperiodic neuroendocrine axis between the two taxa.

LH and FSH plasma levels have similar profiles over the course of a breeding season in several photoperiodic birds (Silverin *et al.*, 1999), but different mechanisms appear to regulate their release. GnRH1 acts upon the gonadotropic cells of the pars distalis to stimulate the release of LH (Sharp *et al.*, 1998), which can happen after a single long day (Nakao *et al.*, 2008). Expression of the GnRH1 receptor varies throughout the reproductive cycle in at least one bird species (Bedecarrats *et al.*, 2006)

but how this may mediate photoperiodic response is unclear. Although FSH has an important role in the stimulation of sex hormones and gonadal function, the direct mechanism of its regulation is unclear, but is unlikely to be GnRH1 in at least one photoperiodic bird (Proudman *et al.*, 2006). Lamprey-GnRH3 regulates FSH in some mammals, but its role in birds, especially in response to photoperiod is unknown (Leska *et al.*, 2007).

Birds are similar to fishes, but different from mammals in that, separate cells in the pars distalis produce FSH and LH (Proudman *et al.*, 1999; Puebla-Osorio *et al.*, 2002). The functional significance of this difference is unknown.

As in mammals, the main sex hormones in birds are testosterone and estradiol. As in mammals and fishes, sex hormone feedback on gonadotropin expression mediates sexual maturation in response to photoperiod (Dawson & Sharp, 2007). Overall, the hormonal and anatomical basis of the photoperiodic neuroendocrine axis is highly conserved between mammals and birds.

Fishes

Signal reception and encoding

In fishes, photoperiod signal reception appears to be extraretinal and extrapineal (Masuda *et al.*, 2005; Borg, 2010). This suggests that deep brain photoreceptors receive the photoperiodic signal, as in birds. Indeed, multiple photoreceptor types are distributed throughout the brain of two photoperiodic fishes: the Atlantic salmon (Philp *et al.*, 2000) and the common minnow (Álvarez-Viejo *et al.*, 2004). Unlike birds, the wavelength of light does not appear to affect photoperiodic inducement of sexual maturation

(McInernev & Evans, 1970). Although these results are based on a single species, they suggest multiple photoreceptors can receive the photoperiod signal and communicate it to the neuroendocrine axis.

The daily melatonin rhythm has been proposed as an alternative to neuronal communication of photoperiod in some fishes (Migaud *et al.*, 2010). However, methodological limitations make it difficult to distinguish between the potential effects of melatonin as a communicator of the photoperiod signal and its effects on targets downstream of reception and initial response to photoperiod (Mayer *et al.*, 1997; Bromage *et al.*, 2001; Borg, 2010). We can conclude that the melatonin signal does not affect signal reception or downstream response in at least some photoperiodic fish (Masuda *et al.*, 2005; Borg, 2010) and further work is required to distinguish between potential effects on these elements in other fishes.

Neuroendocrine axis

Of the three taxa, the least is known about the neuroendocrine axis in photoperiodic fishes. Hormonal response appears to be largely conserved, but neuroanatomy is not, with differences between the mammal, bird and fish neuroanatomy affecting how hormonal signals are transmitted between the brain and pituitary. It must be noted that anatomical regions within the hormone-secreting adenohypophysis of the pituitary are more distinct in mammals and birds relative to fishes. In mammals and birds, the adenohypophysis contains the pars tuberalis, the location of the earliest hormonal response to long days, and the pars distalis, which secretes gonadotropins in response to GnRH stimulation (Kah & Dufour, 2010). These regions are not

morphologically distinct in fishes. To avoid confusion, they will be referred to in fishes by the corresponding mammalian structure when localization of gonadotropin expression makes this possible.

The earliest long day hormonal response in mammals and birds is a TSH increase in the pars tuberalis, which stimulates Dio2 production in the MBH. There is no clear anatomical analog of the pars tuberalis in fishes (Kah & Dufour, 2010), but TSH is produced in the pars distalis of the fish pituitary (Kasper *et al.*, 2006; Cerdá-Reverter & Canosa, 2009). This distribution of TSH and the lack of a median eminence in higher teleosts (Cerdá-Reverter & Canosa, 2009) suggest that if TSH does stimulate *dio2*, it would do so through neuronal communication between the pars distalis and MBH. For now, its photoperiodic response remains unknown.

As in mammals and birds, fish Dio2 catalyzes the conversion of T4 to the bioactive T3 (Orozco & Valverde, 2005). The long day response of *dio2* and its distribution within the brain of photoperiodic fishes is unknown.

The role of TH hormone in photoperiodically induced sexual maturation in fishes is unclear. Methodology and focus (e.g. TH levels in plasma versus TH levels in the brain) often differ among studies, making it difficult to compare results and make general inferences regarding its possible role. In addition, many of these studies did not control for the potential effects of other environmental signals, including temperature and nutrient availability (Raine, 2010). Nevertheless, there are examples of T3 increasing during early sexual maturation in photoperiodic fishes (Cyr & Eales, 1996; Nordberg *et al.*, 2004). In at least one species, a change in the photoperiod regime is the specific trigger for T3 increase (Cyr *et al.*, 1988). In general, there is a correlation between
reproductive stage and TH levels in seasonally breeding fishes (Cyr & Eales, 1996), although the extent to which photoperiod or other environmental signals mediate TH levels in these fishes is unknown. T3 can stimulate GnRH neurons in fish (Parhar *et al.*, 2000), but this has not been tested in photoperiodic fishes.

To date, three GnRH paralogs have been identified in teleosts: GnRH1 and GnRH2 also occur in mammals and birds, but GnRH3 is unique to teleosts (Kah *et al.*, 2007). As in mammals and birds, GnRH1 neurons are found in the preoptic area. They share a common developmental origin with GnRH3 neurons (Kah *et al.*, 2007), which are found primarily in the ventral telencephalon (Cerdá-Reverter & Canosa, 2009) although GnRH3 distribution often overlaps with GnRH1 (Kah *et al.*, 2007). GnRH2 neurons are distributed throughout the midbrain tegmentum (Cerdá-Reverter & Canosa, 2009).

Any addition functions of the three paralogs beyond stimulation of the gonadotropins are unclear. GnRH1 is considered the hypophysiotropic paralog as it is expressed in neurons that innervate the pituitary and is capable of stimulating gonadotropin production and gonadal development (Cerdá-Reverter & Canosa, 2009). In teleosts where GnRH1 is not present, GnRH3 has been shown to be the hypophysiotropic paralog (Chen & Fernald, 2008; Cerdá-Reverter & Canosa, 2009). GnRH2 is thought to regulate sexual behavior as a neuromodulator (Cerdá-Reverter & Canosa, 2009). This is its primary role in mammals and birds, but this conclusion is based on its distribution within the brain and has not been functionally examined.

Work on GnRH expression in photoperiodic fishes is limited. In the masu salmon and rainbow trout, photoperiod treatment stimulates production of GnRH3 (Amano *et al.*, 1995; Davies *et al.*, 1999), but not of GnRH2 (Bromage *et al.*, 2001). Long days increase

expression of all GnRH paralogs in the photoperiodic pejerrey (Miranda *et al.*, 2009). In the photoperiodic grey mullet, GnRH1 increases, but GnRH3 decreases during sexual maturation (Nocillado *et al.*, 2007). These results indicate that the role of GnRH1 as the main hypophysiotropic paralog is largely conserved among mammals, birds, and fishes, but they also support functional differences among fishes during photoperiodic response.

There are two anatomical differences in gonadotropin secretion among the mammal and bird models relative to fishes that may affect response of the fish neuroendocrine axis to photoperiodic stimulation. First, stimulation of gonadotropin secretion by GnRH neurons must be via direct innervation from the hypothalamus to the pituitary because there is no median eminence in higher teleosts. The pars distalis of at least one photoperiodic fish has extensive GnRH innervation (Andersson *et al.*, 1995), but the effect of photoperiod on its expression has not been studied. Second, in mammals the gonadotropins are produced in the same cells (Childs, 2006), whereas they are produced in separate cells in fishes and birds (Puebla-Osorio *et al.*, 2002; Kanda *et al.*, 2011). The significance of this is unknown, but it may reflect differences in regulation of gonadotropin secretion.

Despite these differences and uncertainty, studies in photoperiodic fishes show that GnRH stimulates the gonadotropins, in manner that is largely similar to its actions in photoperiodic mammals and birds (Davies *et al.*, 1999; Amano *et al.*, 2001; Hellqvist *et al.*, 2006; Miranda *et al.*, 2009; Choi *et al.*, 2010). Long day exposure results in gonadotropin stimulation and these gonadotropins are transported from the pars distalis to the gonads where they induce sex hormone production (Borg, 2010). Gonadotropins stimulate steroid hormone production from the gonad, which have feedback effects on gonadotropin secretion. The main sex hormones are estradiol and 11ketotestosterone, although testosterone also has androgenic effects (Borg, 2010).

Section summary

Although the precise mechanisms of photoperiod signal reception and transduction to the neuroendocrine axis have not been established in birds or teleost fishes, all evidence suggests that retinal reception and melatonin communication in mammals are derived traits. Testing this hypothesis requires establishing a teleost model of photoperiodic response in which the techniques necessary to study these mechanisms are practical. Previous work using the threespine stickleback (McInernev & Evans, 1970; Borg, 2010) and data presented herein (Chapter III; Chapter IV) demonstrates that it meets this criterion.

The hormonal and anatomical basis of the photoperiodic neuroendocrine axis is highly conserved between mammals and birds. Available data regarding the function and photoperiodic responses of GnRH and the gonadotropins suggest they are conserved with respect to fishes as well, but there are several important unknowns that make this an open question. First, the functions, locations, and responses of TSH and TH in long days are unknown in fishes. Many studies on photoperiodic fishes do not control for the potential effects of environmental signals beside photoperiod, which has limited our ability to compare results among fishes or to other vertebrate taxa. Furthermore, the lack of a median eminence and differences in pituitary morphology in fishes relative to mammals

and birds have not been examined in the context of the photoperiodic neuroendocrine axis.

Such studies require a teleost fish model that can be manipulated in controlled conditions and is amenable to the techniques whose results can be directly compared to mammals and birds. Chapter III demonstrates that threespine stickleback from multiple populations have a strong photoperiodic response in controlled conditions, which allows phenotypic and physiological results to be interpreted solely in relation the influence of photoperiod. Chapter IV establishes the stickleback as a teleost model for the photoperiodic neuroendocrine axis by measuring gene expression of key hormones in the TH pathway. The findings are discussed in relation to other photoperiodic vertebrates.

Brief outline of this dissertation

Despite the prevalence of photoperiodism within vertebrates, and its obvious ecological significance, important questions regarding its variation with geography and its physiological and genetic foundations remain. These are:

- How do geography and environment affect vertebrate photoperiodism?
- Are the daily circadian clocks and the seasonal photoperiodic timer causally connected?
- Is the physiological basis of photoperiodic response conserved among vertebrates?
- What is the mechanism by which animals interpret photoperiod?
- What is the genetic basis of variation in photoperiodic response?

Progress on these questions has been limited by; (1) a reliance on the candidate gene approach and inappropriate assumptions regarding interpretation of these results, (2) models where consideration of intraspecific variation is impractical or has not been conducted, and (3) models in which we cannot attribute phenotypic changes solely to photoperiodic response and not the correlated effects of other environmental signals. This dissertation addresses these limitations by empirically demonstrating proper methodology for the study of photoperiodic time measurement and by establishing the threespine stickleback as a model of vertebrate photoperiodism.

The threespine stickleback is a small teleost fish with a diversity of phenotypes and life history forms (Bell & Foster, 1994). Long or increasing day lengths stimulate sexual maturation in both sexes (Baggerman, 1985; Yeates-Burghart *et al.*, 2009). Phenological variation has been observed between populations from different latitudes (Borg, 1982; Crivelli & Britton, 1987). This may be due to variation in photoperiodic response, but its contribution cannot be distinguished from the potential effects of other environmental signals. The gonadotropins have an annual cycle in wild-caught individuals, with plasma concentrations peaking early in the breeding season (Hellqvist *et al.*, 2006), which suggests that at least the output of the photoperiodic neuroendocrine axis is conserved among mammals, birds and the threespine stickleback.

Chapter II and part of Chapter III were motivated by a question I've been asked at several conferences: "Shouldn't you be using the circadian clock to understand photoperiodism?" This assumption is particularly widespread in teleost fish biology, where the previously described work in Pacific salmonids is widely accepted. Although this proposed connection is an intuitive solution for control of organismal action on two

time scales, we must consider it an open question because of the untested assumptions described above and the work presented herein.

Chapter II is the result of collaboration between W.E. Bradshaw, C.M. Holzapfel and myself and has been previously published (O'Brien *et al.*, 2011). We address the potential for autocorrelation to produce an assumption of a causal (i.e. necessary) relationship between two variables. Two traits are often assumed to have a causal relationship with one another because both covary with a third factor. We describe a simple method to test for autocorrelation and apply it to the previously described latitudinal clines in Pacific salmon *clock* gene alleles. The results demonstrate that phenological variation and *clock* gene allelic variation are uncorrelated, which means they cannot be causally connected. We suggest this technique as a necessary first step when studying covariation across geography or any phenomena where a causal relationship is being investigated and autocorrelation may exist.

Chapter III is the result of collaboration between L. Unruh, C. Zimmerman, W.E. Bradshaw, C.M. Holzapfel, W.E. Cresko and myself. Threespine stickleback populations from Alaska and Oregon were raised in a common environment and phenotyped for photoperiodic response. We show that all populations are photoperiodic and there is no difference in response within or between the two latitudes. This result was unexpected, as the difference between Alaska and Oregon (~ 18°) is generally large enough to demonstrate intraspecific genetic variation in photoperiodic response or proxies of photoperiodic response (reviewed in Danilevskii, 1965; Bradshaw & Holzapfel, 2007). We suggest that this constancy of response in the lab may be mediated by differences in response to increasing temperature in the wild. These results are the first steps in

establishing the threespine stickleback as a vertebrate model of photoperiodic response. By phenotyping multiple, outbred populations absent of maternal or field effects, we demonstrate the utility of the threespine stickleback for the study of photoperiodic response.

In Chapter III, we also show that there is no partitioning of variation in threespine stickleback *clock* allele length within or between the two latitudes. We discuss these results in the context of the assumption that *clock* allelic variation affects photoperiodic response through its role in the circadian clock. Our data is further evidence that the role of *clock* in seasonal timing is equivocal at best. We urge caution in the interpretation of allelic variation across geographic gradients.

Chapter IV is the result of collaboration between R. Bourdo, W.E. Bradshaw, C.M. Holzapfel, W.E. Cresko and myself. Photoperiodic response of the neuroendocrine axis that initiates sexual maturation is conserved between mammals and birds, but it is unknown if it is observed across vertebrates, including teleost fishes. To fully address this hypothesis, studies comparable in detail and technique to those performed in tetrapods must be conducted in teleost fishes. We quantified the photoperiodic response of genes coding for key hormones in the thyroid hormone pathway in three populations of threespine stickleback. These data were complemented with spatial expression analysis when anatomical location of a hormone's response in fish was unknown. Response of the neuroendocrine axis implicated the thyroid hormone pathway in the photoperiodic initiation of sexual maturation and was robust among the study populations.

This study is the first to examine early photoperiodic response of the neuroendocrine axis in a teleost fish in highly controlled experimental conditions.

Chapter III shows that the phenotypic response to photoperiod does not differ between two latitudes. Herein we demonstrate that the neuroendocrine basis of this response is also consistent. These data strongly support conservation of the photoperiodic neuroendocrine axis among mammals, birds and fishes, which suggests that results from the study of threespine stickleback photoperiodism can be applied to our understanding of vertebrate photoperiodism in general.

Finally, Chapter V summarizes the results from Chapters II – IV, discusses how they have contributed to the study of biological timing and our understanding of vertebrate photoperiodism, and suggest future avenues of research.

CHAPTER II

TESTING FOR CAUSALITY IN COVARYING TRAITS: GENES AND LATITUDE IN A MOLECULAR WORLD

A paper published in *Molecular Ecology*, authored by C. O'Brien, William E. Bradshaw, and Christina M. Holzapfel

With the advent of modern molecular techniques, increasing attention is being paid to non-model organisms for investigating the genetic basis of various phenotypes in physiological, ecological or geographical contexts. As genes are discovered that covary with an environmental parameter such as temperature, light or latitude, there is a natural temptation to ascribe causality to these correlations. However, correlations are only the tantalizing starting points for robust experimental designs and, in themselves provide evidence for neither causality nor an underlying functional mechanism. Herein, we use covariation of traits with latitude to illustrate the problem of confounding causation and correlation over geographic gradients. We begin with a simple diagram:



If A is correlated with C and B is correlated with C, then A will automatically be correlated with B. There follows the natural temptation to infer or conclude that A causes B, that is genetic variation in A constitutes the genetic basis of B. As an example, we consider the relationship between the circadian clock regulating daily activities of organisms and the photoperiodic timer regulating seasonal activities of organisms. This relationship has a long and contentious history (Tauber & Kyriacou, 2001; Hazlerigg & Loudon, 2008; Bradshaw & Holzapfel, 2010; Saunders, 2010; Koštál, 2011), a legacy of Bünning's (1936) proposition that the circadian clock formed the causal basis of photoperiodism. At the molecular level, a probabilistic cause between circadian rhythmicity and photoperiodism occurs in plants (Kobayashi & Weigel, 2007; Wilczek et al., 2009) and in a long-established laboratory strain of Syrian hamsters (Shimomura et al., 1997; Lowrey et al., 2000). However, there are no examples where the circadian clock has been shown to be necessary, let alone sufficient for regulating photoperiodic response in natural populations of any animal. Yet, elements of the circadian clock have been shown to vary with latitude, as have phenotypes of the photoperiodic timing mechanism (Fig. 2.1). Therein lies the problem: Covariation is not proof of causation.



Fig. 2.1. Inference of causality between circadian rhythmicity and photoperiodism due to their common covariation with the independent variable, latitude. If allelic variation in a circadian gene is correlated with latitude and a proxy for the photoperiodic timer is correlated with latitude, the incorrect conclusion could be drawn that the circadian clock forms the causal basis of the photoperiodic timer, that is that the circadian clock is necessary for or forms the mechanistic basis of photoperiodic time measurement. In fact, the circadian clock, the photoperiodic timer, and an endless array of other variables are correlated with latitude but are not necessarily causally connected.

The seasonal timing of life-history events, which is typically orchestrated by the photoperiodic timer, is correlated with latitude in both plants (Wilczek *et al.*, 2009) and animals (Bradshaw & Holzapfel, 2007). An increasing number of circadian-related genes are now known to vary with latitude in *Neurospora* (Michael *et al.*, 2007), plants (*Arabidopsis*: Michael *et al.*, 2003; Caicedo *et al.*, 2004; Stinchcombe *et al.*, 2004; Glycine: Zhang *et al.*, 2008), *Drosophila* (Kyriacou *et al.*, 2008; Rand *et al.*, 2010), fish (O'Malley & Banks, 2008a, O'Malley *et al.*, 2010), birds (Johnsen *et al.*, 2007), and humans (Cruciani *et al.*, 2008). Given the observation that both circadian genes and photoperiodically mediated seasonal traits vary with latitude, the tendency is to conclude a causal connection between the circadian clock and the photoperiodic timer based on their latitudinal covariation.

The covariation of two traits with latitude could indeed be due to a common causal mechanism (pleiotropy), in which case an interesting relationship has been established and the question then becomes resolving the mechanistic basis of their coevolution. However, while latitude usually and appropriately serves as a composite variable, latitudinal variation represents multiple environmental factors, any one or a combination of which could be exerting parallel selective forces. The covariation of two traits, the same selective force acting on two genetically independent traits, or a single selective force acting on one trait accompanied by genetic hitchhiking of a closely linked trait (Li, 1997; Schluter *et al.*, 2004; 2010). Examination of the relationship between variables can be made using techniques described in Sokal and Rohlf (1995): partial correlation

are held constant; path analysis incorporates simultaneously the contribution of several correlated variables. While useful, these statistics are complex, may suffer from collinearity of the independent variables (Petraitis et al., 1996), are not readily accessible in many statistical packages and heretofore have not incorporated discrete variables. We are proposing a more transparent test that requires little more than a hand calculator or an Excel spreadsheet and incorporates both linear regression and analysis of variance. Below, we provide examples from flies and fish to illustrate the simplicity and usefulness of the analysis of residuals to avoid a spurious conclusion of causation when only correlation exists. When Y is regressed on X, the regression equation, $^{A}Y = a + bX$ plots the regression line and $Y_i - ^Y =$ deviations from regression (residuals). The residuals are zero correlated with X, that is the effect of X on Y has been factored out. If A is a causal element of B and both are correlated with latitude, then even when the common element of latitude is factored out the residuals should still be correlated; if not, their common correlation with latitude is due to linkage or independent evolution and not due to a basic underlying causal relationship between A and B. When A or B is a discrete and not a continuous variable, the residuals are computed as deviations from mean latitude for each category of Y. Although applicable to the covariation or association of any two traits or processes with any environmental parameter, we continue with examples from the biological timing literature. To illustrate the test, we have chosen two specific examples because of their connection with latitude, because of the large number of sample populations over a wide latitudinal range, and because the numerical data were available in the source papers. This sort of analysis was not possible for most of the papers we read because either the sample size was too small or the tabular, numerical data from which

figures were generated were not available either in the body of the text or in supplemental online material. The advent of requiring the posting of such data (Fairbairn, 2011) will make subsequent verification via independent analysis tractable.

First, in *Drosophila littoralis*, Lankinen (1986) found significant correlations between latitude and a proxy for the photoperiodic timer (critical photoperiod necessary to induce adult diapause) and between latitude and the two most fundamental properties of any circadian rhythm (the period and amplitude of its oscillation) (Fig. 2.2). Insightfully, he factored out the common effect of latitude and showed that the residuals of critical photoperiod were no longer correlated with the residuals of either period or amplitude of the circadian eclosion rhythm. Hence, he proposed that their covariation with latitude was due to linkage and not a causal relationship between them.



Fig. 2.2. Use of residuals to test for a causal relationship between circadian rhythmicity and photoperiodism in *Drosophila littoralis*. (Top) Latitudinal covariation in photoperiodic response (critical photoperiod) and two fundamental properties of the circadian clock, period and amplitude of the oscillation; (Bottom) lack of correlation between deviations from regression of critical photoperiod, period and amplitude on latitude. Any significant relationship between photoperiodic response and properties of the circadian clock is eliminated when the common element of latitude is factored out (plotted from Table 2 in Lankinen, 1986). Details of analyses are in Appendix 2.1.

To verify this conclusion, Lankinen and Forsman (2006) crossed two extreme populations, allowed free recombination and then imposed selection for nondiapause on short days. The hybrid lines exhibited a more 'southern' photoperiodic response and a more 'northern', circadian-based eclosion rhythm than found in any of Lankinen's original geographic strains (Fig. 2.3), that is the reverse of what would have been expected had the circadian clock been a causal component of photoperiodism. These experiments confirmed Lankinen's earlier conclusion (1986) that when the common effect of latitude was factored out, critical photoperiod was not correlated with either fundamental property of circadian rhythmicity. More generally, Lankinen and Forsman's (2006) experiments confirmed the robustness of testing for a potentially causal connection between two traits by using residuals to factor out their common, correlated element.



Fig. 2.3. Verification of analysis of residuals as a test for a causal relationship between photoperiodism and circadian rhythmicity in *D. littoralis* by response to selection on critical photoperiod and period (s) of the circadian oscillation in *D. littoralis*. A northern and a southern population were hybridized, maintained for eight generations on constant light (L:L) to allow free recombination, selected for nondiapause under short days (L:D = 12:12) for 30 generations, maintained in L:L for 10 generations, and the descendents of a full-sib pair maintained in L:L for a further six generations (plotted from data in Lankinen & Forsman, 2006).

Second, in Chinook salmon, *Oncorhynchus tshawytscha*, O'Malley and Banks (2008) found a significant correlation between latitude and their proxy for the circadian clock (length of the polyglutamine repeat in the gene *OtsClock1b*, hereafter, Poly Q) (Fig. 2.4A). They also found a significant association between latitude and their proxy for the photoperiodic timer (run time = seasonal timing of upstream migration in freshwater) (Fig. 2.4B). O'Malley and Banks (2008, p. 2813) conclude with the suggestion 'that length polymorphisms in *OtsClock1b* may be maintained by selection and reflect an adaptation to ecological factors correlated with latitude, such as the seasonally changing daylength.' After extending their correlative analyses to three more species of salmon (*Oncorhynchus*), O'Malley *et al.*, (2010, p. 3705) state more boldly that the '*Clock* gene is a central component of an endogenous circadian clock that senses changes in photoperiod (day length) and mediates seasonal behaviors.' At the heart of the conclusion is the association between latitude, Poly Q and the timing of migration and spawning. This conclusion makes at least three essential, but untested assumptions.



Fig. 2.4. Latitudinal covariation of mean *OtsClk1b* Poly Q domain length (Poly Q) and run (migration) time in Chinook salmon. (a) $r^2 = \text{coefficient}$ of determination from the regression. The plot is redrawn from data from Table 3 in O'Malley & Banks (2008); n = 40, omitting the single 'W' and undefined 'F' runs as did O'Malley & Banks; (b-c) vertical lines show means; $r^2 = \text{reduction}$ in total sum of squares from one-way ANOVA. Plots and analyses are based on the same data set as in (a). Details of analyses are provided in Appendix S2.2.

First, the conclusion requires that a single genotype (high frequency of the 335 allele and concomitant low frequency of the 359 allele) of the Chinook Poly Q domain is the primary determinant of two different run times, spring and autumn (Fig. 2.4B), even in the same river. This assumption may or may not be true. Second, the conclusion assumes that the salmon specific *OtsClock1b* plays a functional role in salmon circadian rhythmicity. There are two *Clock* paralogs in salmon: *OtsClock1a* and *OtsClock1b*, only the latter of which shows a significant correlation with latitude.

However, the assumption that *OtsClock1b* has the same functional role in salmon as its ortholog in the mammalian circadian clock (Baggs *et al.*, 2009) is untested. Third, the conclusion assumes that there is a causal relationship between the daily circadian clock and the seasonal photoperiodic timer. This assumption is at best contentious (Bradshaw & Holzapfel, 2010; Saunders, 2010; Koštál, 2011) and has not been tested in any fish. There is then a great leap from observing a correlation between latitude and only the *OtsClock1b* paralog and a correlation between latitude and run time or spawning date to concluding that the circadian clock is responsible for the evolution of photoperiodism and, hence, seasonal timing (O'Malley et al., 2010). Strictly for purposes of illustration, we assume the first two of the above three assumptions to be true. We then use Lankinen's (1986) approach of analyzing residuals to test for an association between Poly Q and run time by factoring out the effect of latitude on Poly Q. In this case, Poly Q is a continuous variable and run time is a discrete variable. We therefore calculated the residuals from regression of Poly Q on latitude (Fig. 2.4A) and performed one-way ANOVA of the residuals using run time as treatments. After factoring out the effect of

latitude, run time accounted for a non-significant 7% of the residual variation in Poly Q (Fig. 2.4C). We therefore conclude that there is no basis to infer or suggest a causal relationship between them, either as a direct, independent effect of Poly Q on run time or as an indirect effect of Poly Q on the circadian clock. Further discussion of the adaptive significance of Poly Q in relation to run time is unwarranted, as is any speculation about a potential connection between the circadian clock and the seasonal photoperiodic timer. Future research might well be directed towards determining the function and adaptive significance of *Clock1b* in salmon in the context of the circadian clock itself, much as have other studies in diverse organisms (Yerushalmi & Green, 2009).

Hence, we propose that before inferring a causal relationship in similar cases of covariation of two or more traits with a third physiological or ecological independent variable, that a straightforward analysis of deviations from the common independent variable be used. Absent a significant association, no causal relationship should be inferred or suggested. Even an inference of a causal relationship would be reasonable only if all of the following were true: (i) Variation in each trait is significant correlated with a third common element, in our case, with latitude. (ii) The significant correlation between the two traits persists after the effect of the common element is factored out. (iii) The environmental conditions used to show the correlations in (a) and (b) were in the same organism and determined under the same conditions.

Note that our test accommodates the situation where both the trait and the gene are associated with latitude in the same way. In that case, their latitudinal covariation is due to an environmental factor(s) selecting concomitantly on both the gene and the trait; no correlation between them should persist once the latitude-dependent causal

environmental factor(s) is accounted for. If the relationship between the gene and the trait is due to an underlying causal connection, then a significant correlation between them should persist independently of latitude.

Significant, positive results from analysis of residuals serve as a point of departure for future experiments but, in of themselves, do not substitute for an understanding of the functional connection between genotype and phenotype (Kingsolver & Schemske, 1991; Petraitis *et al.*, 1996; Dalziel *et al.*, 2009; Blackman 2010; Storz & Wheat, 2010). Successful connections between molecular variation and functional phenotypes have been established (but only after additional study) in both model organisms such as *Drosophila* (Schmidt *et al.*, 2008; McKechnie *et al.*, 2010; Paaby *et al.*, 2010) or *Arabidopsis* (Wilczek *et al.*, 2009) and in natural populations of non-model organisms such as the house mosquito, *Culex pipiens* (Labbé *et al.*, 2009), lizards (Rosenblum *et al.*, 2010), and organisms cited by Storz and Wheat (2010) and Dalziel *et al.*, (2009), their Appendix S1, Supporting information), including killifish, butterflies, garter snakes, deer mice, oldfield mice, threespine stickleback, and Darwin's finches.

With the advent of tractable molecular approaches in an increasing number of non-model organisms with interesting physiological or ecological backgrounds, there will be increasing impetus to ascribe an adaptive significance to molecular genetic variation. Because postglacial climate change has established many eco-climatic selection gradients across latitudes in nature, any correlation between molecular variation in SNPs, nonsynonymous substitutions or transcriptional profiles with latitude provides a tempting avenue for concluding an adaptive significance for the observed genetic variation. Instead of proposing untested suggestions or implications because of their inherent plausibility,

investigators should first examine residuals as described herein. If non-significant, further discussion or speculation of the potential adaptive significance of their covariation is not warranted. If significant, then an inferred causal connection can be used as a platform from which to seek a functional connection between genotype, phenotype and, ultimately, fitness.

Bridge

In Chapter II, we described how residuals analysis should be used as an initial test of causality when interpreting the relationship between two factors that both covary with latitude. We apply it to two examples, one of which has been interpreted as evidence that the circadian clock is causally involved in photoperiodism. Analysis of this relationship shows that there is no association once the common effect of latitude is factored out. Thus, further investigation of the circadian clock and its relationship to photoperiodism in this context is inappropriate. In Chapter III, we examine this same gene across two latitudes in the threespine stickleback *Gasterosteus aculeatus* and phenotype multiple the photoperiodic responses of multiple populations of threespine stickleback.

CHAPTER III

GEOGRAPHY OF THE CIRCADIAN GENE *CLOCK* AND PHOTOPERIODIC RESPONSE IN WESTERN NORTH AMERICAN POPULATIONS OF THE THREEPSINE STICKLEBACK

An unpublished paper submitted to the *Journal of Fish Biology*, authored by C. O'Brien, L. Unruh, C. Zimmerman, W. E. Bradshaw, C. M. Holzapfel and W. A. Cresko.

Introduction

Proper timing of seasonal events in the life histories of organisms is a key component of fitness at temperate and polar latitudes. A wide variety of animals use the length of day (photoperiodism) to anticipate and prepare in advance for future seasonal changes (Bradshaw & Holzapfel, 2007a). Over 70 years ago, Erwin Bünning (1936) proposed that the circadian clock that organizes the daily activities of organisms also formed the basis of the seasonal photoperiodic timer. Evidence for this proposition is strongest in plants (Kobayashi & Weigel, 2007; Wilczek *et al.*, 2009) and highly inbred strains of the golden hamster (Shimomura *et al.*, 1997; Lowrey *et al.*, 2000). Otherwise, the connection between the two physiological processes remains highly contentious (Hazlerigg & Loudon, 2008; Goto *et al.*, 2010; Bradshaw & Holzapfel, 2010a,b; Saunders, 2010; Koštál, 2011; Schiesari *et al.*, 2011). Historically, causal connections

between the daily circadian clock and the seasonal photoperiodic timer were inferred from parallel peculiarities of their physiological behavior to exotic light:dark cycles (vaz Nunes & Saunders, 1999; Tauber & Kyriacou, 2001; Goldman, 2001; Saunders, 2002, 2010, 2011).

With the advent of tractable molecular techniques, a common approach to examine the relationship of circadian and photoperiodic timers has been to use circadian clock genes as candidate loci and then to seek a correlation between mutations or knockdowns of those genes and variation in diapause response in photoperiodic insects (Saunders, 1990; Goto *et al.*, 2006; Stehlík *et al.*, 2008; Han & Denlinger, 2009; Ikeno *et al.*, 2010). The expression of diapause involves a neuroendocrine pathway and it is not clear whether variation in diapause response is due to the effect of the circadian clock on photoperiodism, which is the desired result by the authors, or to an individual clock gene somewhere in the neuroendocrine pathway leading to diapause independently of photoperiodism (Bradshaw & Holzapfel, 2007b; Emerson *et al.*, 2009; Bradshaw & Holzapfel, 2010a, 2010b; Schiesari *et al.*, 2011).

Three logical associations have led investigators to ask whether evolution of the photoperiodic timer, especially over latitudinal gradients, is associated with allelic variation in candidate circadian clock genes segregating in natural populations (Tauber *et al.*, 2007; Mathias *et al.*, 2007; Liedvogel *et al.*, 2009; O'Malley *et al.*, 2010). First, photoperiodism is a physiological mechanism for anticipating seasonal change and preparing in advance for that seasonal change. Second, seasonal environments change with latitude. Third, the timing of seasonal activities (phenology) changes with latitude. The speculative leap in logic is then to assume that any correlation between a circadian

clock gene and latitude or phenology implies a causal connection between the circadian clock and photoperiodism.

The canonical circadian gene *clock* has been the focus of several studies seeking to relate variation in C-terminal polyglutamine domain length within the circadian gene *clock* (PolyQ) to infer a role of the circadian clock in photoperiodism. In *Drosophila melanogaster* deletion of two of the three PolyQ domains of *clock* resulted in altered circadian behavior (Darlington *et al.*, 1998). In mice, excision of a glutamine-rich exon also resulted in altered circadian behavior (King *et al.*, 1997). These findings provided the point of departure for studies aimed at correlating variation in PolyQ with latitude (Johnsen *et al.*, 2007) or with seasonal events acting as a presumptive proxy for photoperiodism in nature (Liedvogel *et al.*, 2009; O'Malley & Banks, 2008a; O'Malley *et al.*, 2010). However, correlation does not demonstrate causation (Kingsolver & Schemske, 1991; Petraitis *et al.*, 1996; O'Brien *et al.*, 2011). In fact, none of the aforementioned studies actually determined photoperiodism under controlled conditions free from maternal or field effects.

Herein, we determine variation in PolyQ and in photoperiodic response as measured by sexual maturation of the threespine stickleback *Gasterosteus aculeatus* L. in northwestern North American populations from Oregon and Alaska (18° difference in latitude). *Gasterosteus aculeatus* is found from marine to freshwater habitats (Bell & Foster 1994), shows extensive population-level variation in phenology in natural populations (Borg, 1982; Crivelli & Britton, 1987), and has been shown to be photoperiodic in both wild-caught (Baggerman, 1985; Bornestaf & Borg, 2000) and

laboratory-reared populations (Yeates-Burghart *et al.*, 2009). Among wild-caught fishes from the Baltic Sea (c. 56-59°N), long days promote reproduction in the late spring and early summer (Borg, 1982; Borg & Van Veen, 1982; Borg *et al.*, 2004). In males, sexual maturation is manifest through increased bright body coloration, territoriality, nest building, courtship, and hypertrophy of the kidney to produce spiggin, the glue used for nest construction (Borg, 1982; Borg *et al.*, 2004; Mayer *et al.*, 2004). Kidney hypertrophy is therefore a reliable indicator of sexual maturity in males. In females, sexual maturation is manifest through increased ovarian mass as a consequence of oocyte maturation (Baggerman, 1972, 1985; Bornestaf *et al.*, 2001; Mayer *et al.*, 2004).

Materials and Methods

Photoperiodic response

Northern (Alaskan) stocks were established from Bear Paw Lake (61°37'N, 149°45'W) and Rabbit Slough (61°34' N, 149°15'W). Southern stocks (Oregon) were established from Cushman Slough (43°36'N, 124°2'W) and Eel Creek (43°35'N, 124°11'W). The animals used for these experiments were G₇ (AK), G₁ (Eel Creek, OR), and G₂ (Cushman Slough, OR) outbred descendants of wild-caught individuals. All collection and care of fish conformed to approved animal care protocols.

The experimental fish were produced, hatched and reared using standard protocols (Cresko *et al.*, 2004; Yeates-Burghart *et al.*, 2009). Briefly, experimental fish were reared on a 10L:14D cycle for 11 - 12 months (Alaska fish) or 11 months (Oregon fish). All fish used in the experiment were at least 50 mm standard length (SL), measured from the dorsum of the pre-maxilla to the end of the caudal peduncle. Within each stock,

fish from several parental lines were pooled and split into male-female pairs for the experiments. Experiments were run in light-tight air-cooled cabinets in climate-controlled rooms at 20°C. Aquaria were visually separated and cleaned separately to avoid the possibility of transferring visual or hormonal cues between aquaria. Fish from each population were exposed to six different photoperiod regimes, ranging from 8L:16D to 23L:1D. Fish that died were not replaced. At the end of six weeks, all surviving fish were included in the data set.

To quantify sexual maturation, the ovary-somatic index (I_0) and the kidneysomatic index (I_K) were determined. Kidneys or ovaries were dissected out and transferred to 37°C with the respective soma in a desiccator containing Drierite (www.drierite.com) until there was no decrease in mass between two successive weighings. Ovaries, kidneys and soma were weighed using a Mettler AT261 DeltaRange electronic balance (mt.com). I_0 and I_K were calculated as the ratio of ovary and kidney to total body mass, respectively. I_0 and I_K values were raised by 10³ before log transformation to ensure positive values on a log scale.

Clock polyglutamine domain

Northern (Alaskan) collections were made from Bear Paw Lake, Rabbit Slough, Hidden Lake (60°29'N, 150°16'W), and Anchor River (59°45'N, 151°30'W). Rabbit Slough and Anchor River are populations in oceanic environments, whereas Bear Paw Lake and Hidden Lake are isolated freshwater populations. Southern (Oregon) collections were made from Eel Creek, Winchester Marsh (43°16'N, 124°19'W), Miner Creek (43°20'N, 124°22'W), and the junction of the Smith and Umpqua Rivers (43°43'N,

124°05'W). All fish were collected using unbaited minnow traps, anesthetized in MS-222 (Aquatic Eco-systems) and preserved in 200 proof ethanol. DNA was extracted from caudal fin clips using a MasterPure DNA Purification Kit (Epicentre).

The human *clock* ortholog (Ensembl ID ENSG00000134852) was BLASTed against the threespine stickleback genome (Ensembl) to find the gene *clock*. Reciprocal Best Hit (RBH) analysis was then conducted to ensure that the resulting gene was the only *clock* paralog in the stickleback genome. To do so the putative stickleback ortholog was BLASTed against the human genome. The best match that it returned was reciprocally BLASTed against the stickleback genome to ensure that its best match was stickleback *clock*. As an additional check, syntenic analysis of the genomic regions surrounding the *clock* orthologs was performed. The synteny database detects synteny between a specified genomic region (in this case, the genomic region surrounding stickleback *clock*) and regions from an outgroup genome (the human genome) using automated RBH analysis (Catchen *et al.*, 2009).

All further sequence annotation and analysis used Geneious Pro 4.7.6 software (Invitrogen). The stickleback *clock* gene was annotated by identifying exons using Ensembl's automatic gene annotation (Curwen *et al.*, 2004), and then confirmed by comparing the translated protein against the amino acid sequence of other, annotated paralogs. The polyQ domain was apparent in the reference sequence as a region containing only glutamines and a single arginine.

To sequence the polyQ domain, we used flanking primers: a forward primer (CAGGGAGGTCAAACCCAGAC) located on exon 19 of *clock* and a reverse primer (TACTGTGGTTGGCTGCTGAC) located in the 3' UTR. These primers were designed

using NCBI Primer Design (NCBI). PCR products were amplified in an MJ Research PTC-200 (Applied Biosystems): 95°C three minutes, 32 cycles of 95°C 30s, 60°C 30s, 72°C 60s, single cycle of 72°C 7 minutes. Because of a high degree of heterozygosity, PCR products were not sequenced directly, but instead were cloned into a pCR[®] 4-TOPO[®] vector (Invitrogen) and sequenced using a 3130x Genetic Analyzer (Applied Biosystems). In order to capture variation in polyQ length among alleles within individuals, multiple TOPO clones were sequenced from each individual.

Resulting sequences were translated and the polyQ domain was manually annotated in ten fish from each population. Sequences with low quality scores in the domain were discarded and re-sequenced. The number of glutamines within the polyQ domain was counted and the positions of the arginine within the polyQ domain were recorded.

Analyses

We used Microsoft Excel (Microsoft) for linear and quadratic regressions. For regressions of I_0 or I_K on day length, linear regression was always significant (P < 0.003); in no case did the addition of a quadratic term significantly increase the reduction in total sum of squares. We therefore used linear regression for all analyses. We used JMP IN 4 (Sall *et al.*, 2005) for ANOVAs. In the latter case, we modeled latitude (AK = north vs. OR = south) and day lengths as fixed effects. Variation between populations within latitudes was incorporated into the error term.

Results

Photoperiodic response is very similar across populations at northern and southern latitudes

Sexual maturation in both males and females from both northern and southern latitudes increased with day length (Fig. 3.1). The kidney:somatic index (I_K) depended on day length (Two-way ANOVA: $F_{5,219} = 19.4$; P < 0.001) did not differ between northern and southern males ($F_{1,219} = 2.31$; P = 0.130) and there was no latitude by photoperiod interaction ($F_{5,219} = 0.43$; P = 0.829). The ovary:somatic index (I₀) depended upon day length ($F_{5,230} = 20.26$; P < 0.001) and was higher in southern than northern females ($F_{1,230} = 9.28$; P = 0.023) but there was no significant latitude by photoperiod interaction ($F_{5,230} = 0.91$; P = 0.477). These results show that while sexual maturation increased with day length at both latitudes (Fig. 3.1) photoperiodic response did not differ between northern and southern latitudes (no significant photoperiod by latitude interaction).



Fig. 3.1. Photoperiodic response of male and female threespine stickleback (*Gasterosteus aculeatus*) in Oregon (43.5°N) and Alaska (61.5°N) in western North America. Male response is represented by kidney:body mass ratio (I_K); female response is represented by the ovary:body mass ratio (I_O). Open circles show results from Yeates-Burghart *et al.* (2009); closed circles show results from the present study. Error bars are $\pm 2SE$.

The polyglutamine domain of clock varies across individual stickleback but shows no

population structuring

The BLAST search and syntenic analysis found one *H. sapiens clock* ortholog in the stickleback genome (Ensembl ID ENSGACG00000015939) (Supplementary Fig. 3.1). *Gasterosteous aculeatus clock* contains 20 exons from bp 489,361 – 499,374 on linkage group IX (Ensembl). Examination of the sequence shows that the polyQ domain is located in exon 20.

The polyQ domains (Fig. 3.2A) contained between 22 and 38 glutamine repeats and did not differ between latitudes (Nested ANOVA: $F_{1, 6} = 0.74$; P = 0.422) or among populations within latitudes ($F_{6,72} = 1.162$; P = 0.336). An arginine residue (Fig. 3.2B) occurred within each of the PolyQ domains between positions 2 and 26. Mean position of the arginine residue did not differ between latitudes ($F_{1,6} = 0.533$; P = 0.493) or among populations within latitudes ($F_{6,72} = 0.907$; P = 0.495). These results show that there is no significant difference in either length of the PolyQ domain or position of the arginine residue within the PolyQ domain between latitudes or among populations within the northern (AK) and southern (OR) latitudes.



Fig. 3.2. Polyglutamine domain (PolyQ) in the *clock* gene in southern (Oregon) and northern (Alaska) populations of *G. aculeatus*. (a) Domain length in number of glutamine repeats; (b) position of the arginine codon within the polyglutamine domain. Error bars are $\pm 2SE$.

Discussion

Stickleback have similar photoperiodic responses at northern and southern latitudes

Previously (Yeates-Burghart *et al.*, 2009), found that photoperiodic response of a single southern (Oregon) population exhibited no significant variation with photoperiod in either ovarian development or male kidney enlargement whereas a single northern (Alaska) population exhibited a strong photoperiodic response. After using replicate populations within Oregon and Alaska (Fig. 3.1), it is now clear that threespine stickleback are photoperiodic at both latitudes and do not differ in photoperiodic response between latitudes. This pattern is inconsistent with other vertebrates where photoperiodic response tends to increase with latitude and northern populations typically exhibit a

stronger photoperiodic response than southern populations (Bradshaw & Holzapfel, 2007a). In both Yeates-Burghart *et al.* (2009) and the present study all experiments were run at 20°C using laboratory-reared fishes where field and maternal effects were minimized. Experimental fishes consisted of a single male paired with a single female that were visually and chemically isolated from other experimental fish and, hence, represented independent replicates. Consequently, the similarity in their photoperiodic responses cannot be ascribed to phenotypically plastic responses to a variable environment or to visual or water-borne cues. We therefore conclude that genetically determined photoperiodic responses do not differ between Oregon and Alaskan populations separated by ~18° of latitude.

Constancy of photoperiodic response in a common laboratory environment does not necessarily translate into a constant physiological response to natural environments over a latitudinal gradient. In threespine stickleback from the field, gonadal maturation is accelerated by both increasing day lengths and warmer temperatures (Borg, 1982; Borg *et al.*, 1987; Andersson *et al.*, 1992; Hellqvist *et al.*, 2004) and cold-acclimated fishes have greater facility in adjusting to warm temperatures with increasing day lengths (Guderley *et al.*, 2001). These physiological responses to day length and temperature need to be considered in the context of the photic and thermal environments of Alaska and Oregon. We only manipulated one of these variables, photoperiod, while keeping the others constant. Although climates are colder in coastal Alaska than Oregon (U. S. Department of Commerce, 1968), spring and summer day lengths are longer and spring temperatures rise faster in Alaska than Oregon (Fig. 3.3). We therefore propose that the accelerating effects of longer day lengths and increasing temperatures in the more northern

environment may compensate for the lower average temperature in Alaska than Oregon. Hence, northern fishes would be reproductively prepared to exploit the shorter northern growing season during the brief period when summer waters are warmest. Finally, we encourage rearing animals from different localities in a common environment before using them to infer an underlying genetic basis for differences in functional phenotypes.



Fig. 3.3. Day length and temperature profiles during the winter and spring in Oregon (OR) and Alaska (AK). (a) Circles show the day lengths at which photoperiodic responses were determined. Note that the Oregon populations do not experience day lengths as short as eight hours or as long as 17 hours light per day. Day lengths are calculated as the time from the onset of civil twilight in the dawn until the end of civil twilight in the dusk for Florence, OR, and Seward, AK (http://www.sunrisesunset.com).
(b) Water temperatures in the Rogue River near Agness, OR (42° 34.7' N, USGS 14372300), and Wasilla Creek, near Palmer, AK (61°38.5' N, USGS 15285000), based on data from 2010 and 2011 (http://waterdata.usgs.gov/usa/nwis/).

Absence of clock *polyglutamine domain length (polyQ)*

In *Drosophila melanogaster*, the Clock protein heterodimerizes with the Cycle protein to promote the transcription of period and timeless. Heterodimerization of Period and Timeless and their migration into the nucleus lead to the inhibition of their own transcription by Clock and Cycle (Darlington *et al.*, 1998). The interest in PolyQ comes from the observation that "a truncated dCLOCK protein lacking two of the three polyglutamine repeats [dCLOCK (Δ Q)] only weakly activates per and tim" (Darlington *et al.*

al., 1998, p. 1602). In the mouse, the *clock*^{Δ 19} mutant results in a long circadian period (Gekakis *et al.*, 1998; Jin *et al.*, 1999; Lowrey & Takahashi, 2004). King *et al.* (1997), found that "an AàT transversion at the third base position of the 5' splice donor site of intron 19" results in skipping the exon immediately upstream, i.e., exon 19. Exon 19 is in the "glutamine-rich region of the C-terminus of the predicted Clock protein (amino acids 514-564)," but not in the downstream PolyQ region (amino acids 739-837) (King *et al.*, 1997). These studies provided new and interesting insights into *clock* in the context of daily circadian timing, but they revealed nothing about any relationship between circadian rhythmicity and photoperiodism. The tractability of measuring PolyQ provided a convenient proxy for variation in the circadian clock that potentially could create functional differences in circadian rhythmicity. Unfortunately, various investigators made a logical error by seeking a causative relationship between variation in PolyQ and latitude or phenology as assumed proxies for the photoperiodic timer.

Our findings of a lack of correlation between polyQ domain and aspects of photoperiodic response are not unique. We found no association between PolyQ and latitude in western North American populations of stickleback (Fig. 3.2). Similarly in the European blue throat Luscinia svecica there is no correlation between PolyQ and latitude from Armenia to Norway ($40^{\circ}30^{\circ} - 70^{\circ}30^{\circ}N$) (Johnsen *et al.*, 2007). Hence, in both species, there is no evidence of a connection between PolyQ and local or regional variation in phenology or photoperiodic response.

Photoperiodism, more than any other proximal factor, is responsible for the onset of first clutches among populations of the blue tit *Cyanistes caeruleus* and photoperiodic

response can vary between island and mainland populations at the same latitude (Lambrechts *et al.*, 1997). In a transect from Italy to Finland (36°44' – 62°37'N), Johnsen *et al.* (2007) found a significant correlation between latitude and PolyQ but only when an atypical, monomorphic, southernmost population was entered into the correlation. Johnsen *et al.* (2007) did not provide any correlation between PolyQ variation and phenological events and, in fact, made the appropriate warning (p. 4878): "Determination of the phenotypic effects of different ClkpolyQcds alleles described here would require detailed studies of both circadian and photoperiod-related behaviours of birds of differing ClkpolyQ genotypes."

Within a single site (Wytham Woods, UK; 51°47'N), Liedvogel *et al.* (2009) sought to correlate PolyQ with laying date, hatch date, and incubation duration of 950 blue tits over a two-years period. No "significant overall year*genotype interaction was found for any of the timing traits in focus (all results with P > 0.213)." However, when the authors continued their search for significance within the observed "non-significant" data, they found that by considering the second year in isolation, they could find a significant correlation between PolyQ and laying date and hatch date (P = 0.047 and P = 0.033, respectively, but without any table-wide adjustment for a-posteriori multiple comparisons). A follow up study on a great tit *Parus major* population at the same site found no association between PolyQ and the same measures of reproductive timing (Liedvogel & Sheldon, 2010). Hence, studies among birds over a large latitudinal range or within a single locality with a large sample size provide at best equivocal evidence for an association between *clock* polyglutamine repeat length and the timing of phenological events, much less photoperiodism.

Among teleost fishes, the molecular basis of daily circadian rhythmicity has been studied in the zebrafish *Danio rerio*. In zebrafish, the core loop of the circadian clock involves three paralogs of *clock* whose proteins form heterodimers with three paralogs of bmal that drive rhythmic expression of three paralogs of period and cryptochrome (Vatine *et al.*, 2011). No connection has been made between any core circadian rhythm genes and photoperiodically controlled seasonal life histories in zebrafish.

Salmonids as a family are photoperiodic for many seasonal life-cycle transitions, such as smolting, precocious sexual maturation, migration to sea, and the initiation of migration back to freshwater (Bromage et al., 2001). Two paralogs of clock have been identified in Chinook salmon Oncorhyncus tshawytscha, OtsClock1a and OtsClock1b that arose from a tetraploidation event during divergence of salmonids from other teleost fishes (O'Malley & Banks, 2008b). No functional connection has yet been made between either of these paralogs and circadian rhythmicity in salmonids. Likewise, their functional role in photoperiodism, if any, has not been established. There is no evidence for polyglutamine length polymorphism in the OtsClock1a paralog among four species in the genus Oncorhyncus. In the OtsClock1b paralog, polyglutamine length is polymorphic within and among populations of Chinook, chum O. kita, coho O. kisutch and pink O. gorbuscha salmon (O'Malley & Banks, 2008a; O'Malley et al., 2010). Mean length of the glutamine domain (PolyQ) is not significantly correlated with latitude among 19 populations of coho or 16 populations of pink salmon, but is correlated with latitude in Chinook and chum salmon (O'Malley & Banks, 2008a; O'Malley et al., 2010). O'Malley et al. (2010) used univariate regression trees to identify correlations between the frequency of the most common polyglutamine domain length allele of *OtsClock1b* and

day length on the date of peak spawn and a freshwater migration index over a wide latitudinal range of Chinook, coho, chum and pink salmon. They found that the ability of the univariate regression tree "to assign populations to groups correctly on the basis of these factors" (day lengh and migration index) was not significant (O'Malley *et al.*, 2010, p. 3711) and significant (P < 0.05) only in pink salmon where length of the most common allele varied with day length on the date of peak spawn but not the freshwater migration index. They did not test for a persistent correlation between the frequency of most common *OtsClock1b* allele and latitude after their common covariation with latitude was factored out (O'Brien *et al.*, 2011).

Hence, in fishes as in birds, there is little evidence for a correlation between polyglutamine domain length and latitude or the timing of phenological events. Even if there had been a general pattern of correlation, correlation is not causation (Kingsolver & Schemske, 1991; Petraitis *et al.*, 1996; O'Brien *et al.*, 2011). In neither the birds nor the fishes was there any determination of the actual effect of PolyQ on circadian function or any actual direct measurement of photoperiodic response.

Bridge

In Chapter III, we examine the relationship between allelic variation in a circadian clock gene and latitude, which has been interpreted as evidence that the circadian clock is causally involved in photoperiodism. We show a lack of association with latitude or photoperiodic response in the threespine stickleback. These data demonstrate that caution should be employed when studying genetic variation across ecogeographic gradients and the importance of examining photoperiodism in controlled conditions. Chapter IV builds

on the phenotyping results of Chapter III by examining the physiological basis of photoperiodic response in threespine stickleback manipulated in controlled conditions.
CHAPTER IV

CONSERVATION OF THE PHOTOPERIODIC NEUROENDOCRINE AXIS AMONG VERTEBRATES: EVIDENCE FROM THE TELEOST FISH, *GASTEROSTEUS ACULEATUS*

An unpublished paper submitted to *General and Comparative Endocrinology*, authored by C. O'Brien, R. Bourdo, W. E. Bradshaw, C. M. Holzapfel, and W. A. Cresko.

Introduction

Proper timing of life-history events is critical to fitness (Bradshaw *et al.*, 2004). Photoperiod, or length of day, has a highly reliable annual cycle that makes it an ideal environmental signal that organisms can use to anticipate and prepare for seasonal changes. The use of photoperiod for the timing of sexual maturation and reproduction is widespread among polar and temperate animals (Bradshaw *et al.*, 2007; Goldman *et al.*, 2004). The extensive use of photoperiod across diverse organisms in order to time critical life-history events underscores the importance of this environmental signal for fitness, and leads to the hypothesis that organismal systems that sense and respond to photoperiod have been molded by the action of natural selection for millennia.

Photoperiodic induction of sexual maturation in vertebrate animals begins with reception of a stimulatory photoperiod regime that leads to induction of gonadotropin release, which in turn stimulates production of gonadal sex hormones Bradshaw et al.,

2010). In photoperiodic mammals and birds, the thyroid hormone (TH) pathway initiates the release of gonadotropins (Anisimova & Gascuel, 2006; Nakao *et al.*, 2008; Yasuo & Yoshimura, 2009; Yoshimura, 2010). Although the mechanisms of photoperiod signal reception and transduction differ between mammals and birds, the initial hormonal cascade and its location within the brain are conserved (Fig. 4.1).

Fig. 4.1. (next page). The photoperiodic TH pathway as inferred in fish from mammals and birds. Solid lines and borders indicate established steps and known neuroanatomical locations, respectively. Dashed lines and borders indicate suggested or inferred steps and neuroanatomical locations. Signal reception in mammals and birds: (A) In mammals, the photoperiod signal is received by the retina and neuronally communicated via the suprachiasmatic nuclei (SCN) and superior cervical ganglion (SCG) to the pineal gland (Moore, 1995). Melatonin (mel) produced by the pineal encodes the signal (Cassone, 1998). (B) In birds, the signal is received by extraretinal photoreceptors, most likely hypothalamic opsins (Halford et al., 2009; Nakane, 2010). Communication of the signal is neuronal and does not involve melatonin (Sharp, 2005). The neuroendocrine response in mammals and birds: In both (A) mammals and (B) birds, the earliest hormonal response to a stimulatory photoperiod occurs in the pars tuberalis, where production of thyroid stimulating hormone beta (TSH β) and chorionic gonadotropin alpha (CG α) increase (Hanon et al., 2008; Nakao et al., 2008; Yasuo et al., 2010). TSHB and CGa heterodimerize to form thyroid-stimulating hormone (TSH), which stimulates deiodinase 2 (*dio2*) production in tanycytes lining the third ventricle of the hypothalamus (III-V). Dio2 catalyzes the conversion of the thyroid hormone thyroxin (T4) to the bioactive triiodothyronine (T3) (Hanon et al., 2008; Yoshimura et al., 2010). T3 stimulates production of gonadotropin releasing hormone (GnRH) from neurons in the mediobasal hypothalamus (MBH). GnRH is transported via the pituitary portal system to the pars tuberalis, where it stimulates production of the gonadotropins follicle stimulating hormone (FSHB) and luteinizing hormone (LHB) (Yasuo & Yoshimura, 2009). These heterodimerize with $CG\alpha$ and are released into the bloodstream where they act upon the gonads and other peripheral targets. Signal reception and neuroendocrine response in fish: (C) In fish, photoperiodic signal reception is extraretinal (Borg, 2010; Masuda et al., 2005) and may be an hypothalamic opsin (Philp et al., 2000). The early responses of TSHB and dio2 to a stimulatory photoperiod have not been studied in fish. In general, plasma T3 increases during sexual maturation in photoperiodic fishes (Biswas et al., 2006; Norberg et al., 2004), but its effects on GnRH in photoperiodic fishes are unknown. Photoperiodic manipulation stimulates GnRH, LH and FSH production (Amano et al., 1999; Choi et al., 2010; Hellqvist et al., 2006; Miranda et al., 2009). LH and FSH stimulate the gonads to produce sex hormones (Borg, 2010).



These findings lead to the hypothesis that involvement of TH in reproduction is conserved among vertebrates and therefore may have originated in chordates prior to the diversification of vertebrates (Heyland *et al.*, 2005; Paris *et al.*, 2010). Support for this hypothesis is limited, however, because most studies have occurred on organisms from just the tetrapod clade of vertebrates. A further test of this hypothesis requires comparable studies in other vertebrates, particularly teleost fishes, which is the most speciose vertebrate clade but for which we have precious little data regarding this hypothesis.

Photoperiodic control of sexual maturation is widespread among teleost fishes () (Bromage *et al.*, 2001; Bradshaw & Holzapfel, 2007; Borg, 2010) and the function of the TH pathway is conserved in fish (Orozco & Valverde, 2005; Raine, 2010). However, photoperiodic control of the TH pathway remains unclear (Fig. 4.1C). Our current understanding is limited by an inability to compare studies directly due to differences in measurement techniques, the variety of species examined, and the ability to relate hormonal changes solely to photoperiodic response.

In seasonally reproducing fishes, the bioactive form of thyroid hormone, triiodothyronine (T3), tends to increase during early sexual maturation (Cyr & Eales, 1996; Norberg *et al.*, 2004; Biswas *et al.*, 2006) and, in at least rainbow trout, photoperiod is the specific trigger for this increase in T3 (Cyr *et al.*, 1988). T3 can stimulate GnRH secretion from GnRH neurons in the Nile tilapia (Parhar *et al.*, 2000), but this stimulation of GnRH has not been tested in a photoperiodic fish.

GnRH orthologs are often referred to by the species in which they were first discovered, but can also be referenced by their paralog name to facilitate comparison of their roles among species. We adopted the latter convention for our work. GnRH1 is expressed in neurons located in the preoptic area and GnRH2 neurons are found in the midbrain tegmentum. GnRH3 is unique to teleosts and is expressed in the ventral telencephalon (Chen & Fernald, 2008; Cerdá-Reverter & Canosa, 2009). GnRH1 is considered the hypophysiotropic form, (acting on the pituitary), as it is capable of stimulating gonadotropin production and gonadal development and is expressed in neurons that innervate the pituitary (Cerdá-Reverter & Canosa, 2009). In fishes where GnRH1 is not present, GnRH3 is the hypophysiotropic form (Chen & Fernald, 2008; Cerdá-Reverter & Canosa, 2009). In masu salmon exposed to a stimulatory photoperiod

regime, GnRH3 neurons increase in number (Amano *et al.*, 1999), but the response of GnRH3 in other photoperiodic fishes is unknown.

Extending the role of the TH pathway in photoperiodic induction of sexual maturation to teleosts requires a species with a strong photoperiodic response that can be manipulated in controlled conditions and that can be measured using techniques that make the results comparable to those in mammals, birds and other fishes. These criteria are met in the threespine stickleback, *Gasterosteus aculeatus*, in which we are able to isolate hormonal responses to photoperiod from other environmental variables using controlled laboratory experiments.

The threespine stickleback is a small teleost fish with a wide latitudinal and environmental range that uses photoperiod to initiate sexual maturation (Baggerman, 1985; Borg *et al.*, 2004; Hellqvist *et al.*, 2006; Yeates-Burghart *et al.*, 2009). Like birds (Dawson, 2002; Nakane *et al.*, 2010) and other fishes (Borg, 2010), reception of light related to photoperiodism is extraretinal and extrapineal (Borg *et al.*, 2004). A stimulatory photoperiod increases gonadotropin production (Hellqvist *et al.*, 2004) and wild-caught sticklebacks have an annual cycle of gonadotropin production that peaks early in the reproductive season (Hellqvist *et al.*, 2006). As in mammals and birds, androgens exert a feedback effect on gonadotropin production in both stimulatory and non-stimulatory photoperiod regimes (Borg *et al.*, 2004). In controlled photoperiod conditions in the laboratory, morphological changes of photoperiodic response can be measured Yeates-Burghart *et al.*, 2009; O'Brien *et al.*, in prep).

The goals of this study were to determine if the TH pathway is involved in the photoperiodic initiation of teleost sexual maturation and, if so, whether the dynamics of

the response of the pathway are conserved among mammals, birds and teleosts. To accomplish these goals, we quantified gene expression levels of key TH pathway genes in the brains and pituitaries of threespine stickleback during exposure to a stimulatory photoperiod regime. In the mammal and bird models, an increase in thyroid stimulating hormone (TSH) is the first known response of the photoperiodic neuroendocrine cascade (Fig. 4.1A and Fig. 4.1B). An increase in hypophysiotropic gonadotropin releasing hormone (GnRH) is the first indicator of the initiation of sexual maturation. Luteinizing hormone (LH) is one of the two gonadotropins that are secreted by the pituitary into circulation to stimulate sex hormone production (Fig. 4.1). By measuring these hormones in controlled conditions we determined the effects of photoperiod on expression of these genes independently of other environmental factors. We were then able to make direct comparisons between *G. aculeatus* and mammals and birds. In addition, evaluating multiple populations allowed us to determine the robustness of the results within a single species.

Materials and Methods

Gasterosteus aculeatus stocks

Two northern populations were established from Alaska, Rabbit Slough (AK1: 61°34' N, 149°15'W) and Boot Lake (AK2: 61°43'N, 149°7'W). One southern population was established from Oregon, Eel Creek (OR: 43°35'N, 124°11'W). We will refer to these populations as AK1, AK2, and OR, respectively, throughout the rest of the text.

Crosses were made via *in vitro* fertilization using established laboratory procedures, and then the fish were reared under standard laboratory conditions [Yeates-

Burghart et al., 2009;

stickleback.uoregon.edu/index.php/Crossing_and_Rearing_Protocols]. Experimental fish were reared to adulthood at 20°C on a non-stimulatory 10L:14D photoperiod cycle for 11 – 12 months (L:D = Light:Dark). They were at least 50 mm standard length (SL), as measured from the dorsum of the pre-maxilla to the caudal peduncle before they were subjected to any experimental treatment. All fish care and experimental procedures complied with University of Oregon IACUC-approved animal care protocols.

Experimental design

Conditions were identical to those previously used in measure the phenotypic effects of photoperiod (Yeates-Burghart *et al.*, 2009). One adult male and one adult female were paired in a single aquarium that was visually separated from other aquaria to avoid confounding visual cues. Aquaria were cleaned separately to avoid the possibility of transferring hormonal cues. The fish were fed twice a day *ad libitum*. All experiments were run in light-tight, air-cooled cabinets at 20°C. Photoperiods were programmed with Chrontrol XT electronic timers (<u>www.chrontrol.com</u>).

Upon being placed in the experimental aquaria, male-female pairs were given two short-day cycles of light:dark = 10:14 (hereafter: 10L:14D) before being exposed to 17L:7D stimulatory long days. This long-day regimen was chosen because it is the shortest photoperiod at which phenotypic indicators of sexual maturation in threespine stickleback plateau (Yeates-Burghart *et al.*, 2009). We used the males exclusively for all of the following experiments.

For *in-situ* mRNA hybridization of TSHβ, male stickleback from the AK2 line

were sampled six hours after dawn during a short-day regimen or six hours after dawn after exposure to a single 17L:7D long day regimen. Fish were anesthetized in MS-222 (Sigma) and the entire brain, including the pituitary, was dissected out and stored in 4% paraformaldehyde solution (Sigma Aldrich) at 4°C. The brains with pituitaries were then cryostat sectioned along the coronal plane, and placed on slides that were stored at -80° C until use.

For quantitative real-time PCR measurement of target genes, males from the three populations were sampled six hours after dawn following exposure to 0, 1, 2, 5 or 10 long days. Fish were anesthetized in MS-222 (Sigma) and the entire brain including the pituitary was dissected out and stored in Trizol (Invitrogen) at -80° C. Total RNA was extracted following a standard phenol chloroform protocol. Synthesis of cDNA was performed using random hexamers (Invitrogen) and SuperScript III (Invitrogen). Sample sizes per treatment ranged from 8 - 13 adult males (Supplementary Table 4.1).

Target gene identification

Thyroid stimulating hormone (TSH) and luteinizing hormone (LH) are both heterodimers consisting of a protein-specific β subunit and an α subunit common to TSH and LH. Therefore, we targeted the β subunits to ensure hormone specificity. First, *Homo sapiens and* zebrafish *Danio rerio* TSH β , GnRH and LH β orthologs were compared to the stickleback genome using BLAST to produce a set of candidate genes for further analysis.

Second, we performed phylogenetic reconstructions of the gene families to confirm the identity of the candidate genes, and in the cases of TSHβ and GnRH3, we

annotated all paralogs found in the stickleback genome. Complete amino acid sequences of orthologs from the three gene families were downloaded from the NCBI protein database. If a species contained multiple paralogs, all were included. Alignments of the three gene families were made using Muscle (Edgar, 2004). PhyML 3.0 (Guindon *et al.*, 2009) was used to estimate phylogenies and compute their likelihood scores. The parameters of the phylogenetic model were searched and optimized using M3L (code.google.com/m3l/), which implements the Broyden-Fletcher-Goldfarb-Shanno algorithm (Nocedal, 1980). The best-fitting model for each gene family was selected using the Akaike information criterion test (Akaike, 1973). Approximate likelihood ratio tests for each node were scaled using Shimodaira and Hasegawa (SH-like) support (Anisimova & Gascuel, 2006). For the phylogenetic reconstruction of all three gene families, the best model was JTT (Jones *et al.*, 1992) with a gamma-distributed set of evolutionary rates (Yang, 1996) (Supplementary Fig. 4.1).

Third, we used syntenic analysis of the target orthologs to confirm the results of the phylogenetic reconstructions. The Synteny Database uses Reciprocal Best Hit Analysis (RBH) to detect synteny between two genomes (Catchen *et al.*, 2009; Catchen *et al.*, 2011). Here, genes from a target genome and an outgroup genome are compared to one another using BLAST. Genes in the two genomes are considered orthologous if they are each other's best BLAST matches. If regions of the two genomes have a high number of orthologs, syntenic conservation due to common descent is inferred (Catchen *et al.*, 2009; Catchen *et al.*, 2011). We compared the regions around our target orthologs in the stickleback genome to the spotted green puffer fish *Tetraodon nigroviridis* and *Homo sapiens* genomes using the Synteny Database (Catchen *et al.*, 2011).

mRNA in-situ hybridization

A riboprobe complementary to stickleback TSH β 1 mRNA was synthesized using digoxigenin-labeled UTP (Roche Applied Science). It was hybridized to coronal sections of brains and pituitaries removed from males from the AK2 population to visualize the location of the TSH β expression. The hybridization protocol was adopted from Thisse *et al.* (1993) with the following modifications: sections were not dehydrated prior to hybridization, and incubation with the anti-digoxigenin antiserum solution was performed at room temperature.

Quantitative real-time PCR (qPCR)

First, target and housekeeping gene primer sets were tested using serial dilutions of cDNA to ensure specificity and consistent amplification across a wide range of concentrations. CDNA concentrations of biological samples were quantified using a Qubit fluorometer (Invitrogen). Two hundred nanograms of cDNA were added to individual qPCR reactions. Reactions were performed in 10 µl volumes using a Kapa SYBR ® Fast kit (Kapa Biosystems). Three technical replicates were performed per gene per biological sample.

Two normalization steps created $\Delta\Delta$ Ct values. First, Ct values for technical replicates were averaged and normalized to expression of the housekeeping gene β -actin (Ensembl ID# ENSGACG0000007836) (Hibbeler *et al.*, 2008). The resulting value was then normalized to the Day 0 population means for each individual gene.

Data were analyzed in R using a two-way population*photoperiod treatment with a

Tukey HSD correction for multiple comparisons (R Development Core Team, 2007) and Dunnett's test for comparison of treatment means with a control (Zar, 1996). Both photoperiod and population were treated as fixed effects.

Results

The genomic location and annotation of hypothesized targets of photoperiodism.

We identified two TSH β paralogs in the stickleback genome, confirming previous results (Kitano et al., 2010). TSHB1 (ENSGACG0000005276) is on linkage group XVII and TSHβ2 (ENSGACG0000009897) is on linkage group XII (Fig. 4.2). Phylogenetic reconstruction places them within their expected clades with high support (Supplementary Fig. 4.1A). TSH β 2 is nested within the teleost TSH β 1 clade with the Siberian sturgeon Acipenser baerii, the as an immediate outgroup to the clade containing both TSH β paralogs (Supplementary Fig. 4.1A). This topology indicates that the TSH β duplication resulted from the teleost-specific genome duplication, as the sturgeon lineage is known to have diverged prior to the teleost-specific genome duplication (Postlethwait et al., 2004). Furthermore, there is a high number of paralogs between the genomic regions where TSH β 1 and TSH β 2 are found, indicating that they originated from a single chromosomal region (Fig. 4.2D). As TSH_β1 is the most conserved paralog among teleosts (Fig. 4.2 and Supplementary Fig. 4.1A), and we could not detect TSH_β2 expression in our biological samples, only the photoperiodic response of TSH^β1 was measured.



Fig. 4.2. The threespine stickleback genome contains two thyroid stimulating hormone beta subunit (TSH β) paralogs. Genes along the x-axis and their orthologs are labeled with grey and red dots respectively. Blue circles indicate TSH β orthologs. **(A)** Threespine stickleback TSH β 1 (ENSGACG0000005276), orthologous to the single TSH β in the green spotted puffer fish *Tetraodon nigrovirdis* (ENSTNIG00000018284). **(B)** Threespine stickleback TSH β 2 (ENSGACG00000009897), orthologous to the single *T. nigrovirdis* TSH β . **(C)** *H. sapiens* TSH β (ENSG0000134200), orthologous to the two threespine stickleback TSH β paralogs. **(D)** Syntenic relationships within the threespine stickleback genome show that linkage groups XVII and XII have a high number of paralogs. TSH β 1 (ENSGACG00000005276) and TSH β 2 (ENSGACG00000009897) are labeled.

Two GnRH paralogs (GnRH2 and GnRH3) were found in the stickleback genome, but GnRH1 is absent. GnRH2 (ENSGACG0000009021) is located on linkage group XVII and GnRH3 (ENSGACG0000009582) is on linkage group VI (Fig. 4.3). The phylogenetic reconstruction shows strong support for separation between the three GnRH paralog clades, with the stickleback GnRH paralogs placed in their expected clades (Supplementary Fig. 4.1B). GnRH3 is unique to teleosts and nested within the GnRH1 clade, confirming previous results (Kitano *et al.*, 2010). As GnRH3 is the hypophysiotropic form in fish when GnRH1 is absent (Chen & Fernald, 2008; Cerdá-Reverter, & Canosa, 2009), the photoperiodic response of GnRH3 was measured.



Fig. 4.3. The threespine stickleback genome contains two of the three vertebrate gonadotropin releasing hormone (GnRH) orthologs. Genes along the x-axis and their orthologs are labeled with grey and red dots respectively. Blue circles indicate GnRH orthologs. Grey arrows indicate the expected positions of missing orthologs. (A) Threespine stickleback GnRH2 (ENSGACG0000009021), which has no ortholog in the *H. sapiens* genome. The two megabase region surrounding threespine stickleback GnRH2 has a single ortholog in the *T. nigrovirdis* genome (ENSTNIG0000002767). (C) Threespine stickleback GnRH3 (ENSGACG00000009582), which has no ortholog in the *H. sapiens* genome. The two megabase region stickleback GnRH3 is isolated to show syntenic conservation, but the absence of an ortholog. Threespine stickleback GnRH3 (ENSGACG00000009582), which has no ortholog in the *H. sapiens* genome. The two megabase region stickleback GnRH3 is isolated to show syntenic conservation in the *H. sapiens* genome. The two megabase region stickleback GnRH3 (ENSGACG00000009582), which has no ortholog in the *H. sapiens* genome. The two megabase region surrounding Threespine stickleback GnRH3 is isolated to show syntenic conservation, but the absence of an ortholog. (D) Threespine stickleback GnRH3 has a single ortholog in the T. *nigrovirdis* genome (ENSTNIG00000013337).

A single LHβ ortholog (ENSGACG00000011475) was identified in the

stickleback genome, on linkage group VI (Fig. 4.4 and Supplementary Fig. 4.1C). The

phylogenetic reconstruction shows strong support for separation between the teleost and

tetrapod clades with the LHB clade, with stickleback LHB placed in the expected clades.

Interestingly, synteny of the surrounding genomic region is conserved between stickleback and the spotted green pufferfuish, *T. nigroviridis* (Fig. 4.4B), but not between stickleback and *H. sapiens* (Fig. 4.4A), that the genomic location of LHβ changed after the divergence of teleosts and tetrapods, but prior to the divergence of stickleback and *T. nigrovirdis* from their most recent common ancestor.



Fig. 4.4. The threespine stickleback genome contains a single luteinizing hormone beta subunit (LH β). Genes along the x-axis and their orthologs are labeled with grey and red dots respectively. Blue circles indicate LH β orthologs. Synteny dot plots for threespine stickleback LH β (ENSGACG00000011475), which has a single ortholog in (A) *H. sapiens* (ENSG00000104826) and (B) *T. nigrovirdis* (ENSTNIG0000009862).

TSH β 1 expression is localized to expected regions of the brain

TSH β 1 mRNA is expressed in the pars distalis of the pituitary, as measured by visual inspection of brain section slides after *in situ* hybridization (Fig. 4.5). TSH β 1 may also be expressed around the third ventricle (III-V in Fig. 4.5), although the latter is too faint to distinguish from background staining with certainty. We expected expression to be localized to the pars distalis, as it is the region of the teleost pituitary that produces the gonadotropins (Kah & Dufour, 2010). TSH β 1 mRNA expression appeared to increase after exposure to a single long day (Fig. 4.5).



Fig. 4.5. TSH β 1 expression in the pars distalis of the pituitary labeled via mRNA *in-situ* hybridization. Exemplar coronal sections of the ventral hypothalamus and pituitary from adult male threespine stickleback exposed to (A) short days and (b) one long day. Scale bar is 100 µm. III-V: Third ventricle; PD: pars distalis.

Quantitative real-time PCR (qPCR) shows rapid response of the target genes to

stimulatory photoperiods

ΔΔCt treatment means for TSHβ1, GnRH3 and LHβ are illustrated in Fig. 4.6. Results of the two-way ANOVA are reported in Table 4.1. Photoperiod has a significant effect on the expression of all three genes (for all three, P < 0.0001). There is a significant difference among the populations for GnRH3 (P = 0.001) and LHβ expression (P = 0.015). There is only a photoperiod*population interaction term for TSHβ1 expression (P = 0.037). The significant interaction term for TSHβ1 requires a closer examination of the main effect of photoperiod. As can be seen in Fig. 4.6, the general trend of the effect of photoperiod is still clear across populations, and the significant interaction term is due to a lower level of expression on day 1.

There is a significant difference among the three populations in the response of TSH β 1 to photoperiod. A single long day causes a pulse in TSH β 1 expression in AK1 and AK2, but the response of OR is not significantly different from baseline values (Figure 4.6 and Table 4.1; photoperiod*population effect: *P* = 0.037). This pulse demonstrates a significant effect of photoperiod on TSH β 1 expression (Fig. 4.6 and Table

4.1; photoperiod effect: *P* << 0.0001), although subsequent long days produce no response that is significantly different from baseline values in any of the populations (Fig. 4.6).

		Effect	DF	F-ratio	Р
		Photoperiod	4, 131	25.52	<< .0001
	ΤՏΗβ1	Population	2, 131	1.37	0.26
		Photoperiod * Population	8, 131	2.14	0.037
		Photoperiod	4, 131	21.32	<< .0001
Gene	GnRH3	Population	2, 131	7.09	.0012
		Photoperiod * Population	8, 131	1.62	0.124
		Photoperiod	4, 131	38.89	<< .0001
	LHβ	Population	2, 131	4.35	0.015
		Photoperiod * Population	8, 131	1.74	0.096

Table 4.1. A two-way population*photoperiod ANOVA for TSH β 1, GnRH3, and LH β expression in the brain and pituitary.

There is no difference among the three populations in response of GnRH3 to photoperiod (Fig. 4.6 and Table 4.1; photoperiod effect: P = 0.124). Long days cause a gradual decrease in GnRH3 expression in the brain and pituitary of all populations, with an eventual return to baseline values after six to ten long days (Fig. 4.6 and Table 4.1; photoperiod effect: $P \ll 0.0001$). This decrease is first significant after two to five long days, and the return to baseline levels occurs after five to ten long days (Fig. 4.6). There are significant differences among the populations in overall GnRH3 expression (Fig. 4.6 and Table 4.1; and Table 4.1; population effect: P = 0.00119).

Long days cause a gradual increase in LH β expression in the brain and pituitary of all populations (Fig. 4.6 and Table 4.1; photoperiod effect: *P* << 0.0001). Differences in the timing of this increase among the populations are not significant (Table 4.1; photoperiod*population: *P* = 0.096). There are significant differences among the populations in overall LH β expression (Figure 6; population effect: *P* = 0.015).

Discussion

Answers to primary questions

The primary questions addressed in this study were to ask (1) whether photoperiodic control of sexual maturation occurred via the thyroid hormone (TH) pathway in a teleost fish and, hence, whether this inductive pathway was conserved from fishes to birds and mammals, (2) whether the order of hormonal events in this pathway coincided with birds and mammals, (3) whether this order of events was robust among different populations. In the threespine stickleback, *Gasterosteus aculeatus*, the answer to all three questions is affirmative, but with variations.

Thyroid stimulating hormone

The cascade of response to gonad-stimulating long days begins with up-regulation of thyroid stimulating hormone (TSH). One day of exposure to gonad-stimulating long days elicits an increase in TSH β 1 in the pars distalis of the pituitary (Table 4.1, Fig. 4.5, and Fig. 4.6). In mammals and birds, the first short night elicits a similar response of TSH β (Nakao *et al.*, 2008; Dardente *et al.*, 2010; Masumoto *et al.*, 2010) but in the pars tuberalis of the pituitary (Nakao *et al.*, 2008; Yasuo *et al.*, 2010). As the pars tuberalis as

a distinct region of the pituitary is found only in tetrapods (Kah & Dufour, 2010), the pulse of TSH β 1 in the stickleblack pars distalis indicates that function of the tetrapod pars tuberalis is contained within the pars distalis of teleosts.



Fig. 4.6. The effect of photoperiod on the thyroid stimulating hormone pathway in the brain of adult male threespine stickleback, *Gasterosteous aculeatus*. Male sticklebacks from two populations in Alaska (AK1, AK2) and one population in Oregon (OR) were reared from hatch to adulthood on short days and then exposed to another short day (control, Day 0) or 1, 2, 5, or 10 long days. Expression of TSH β 1, GnRH3 and LH β were quantified with qPCR and long-day treatments normalized to the short-day control. Error bars are ± 2S.E. Sample sizes are given in Supplementary Table 4.1. Time points that share a letter are not significantly different at *P* < .05 according to one-way ANOVA with a Tukey HSD correction for multiple a posteriori comparisons; open bars indicate hormone expression levels that differ significantly from the control at *P* < 0.05 according to Dunnett's test.

A previous study on the stickleback TSH β paralogs found no difference in TSH β 1 pituitary expression between immature and sexually mature adults (Kitano *et al.*, 2010). Our findings contrast with those results (Fig. 4.6), but the studies are not directly comparable. Whereas our study compared lab-raised adult males, Kitano *et al.* (2010) compared TSH β 1 in lab-raised 8-month-old fish on a short day regimen to 12-month-old fish on a long-day regimen, potentially confounding age and time of exposure to long days. The important early pulse of TSH β 1 (Fig. 4.6) would not have been observed by Kitano *et al.* (2010).

The early pulse TSH β 1 was higher in the two more northern populations where the growing season is shorter and the winters are longer and more severe (Table 4.1 and Fig. 4.6). Stickleback in populations that are ecologically similar and geographically proximal to the northern sites in this study breed strictly from mid-May through July (Karvé *et al.*, 2008), whereas in the southern population where winters are mild (O'Brien, in prep.), some sexually mature individuals are found nearly year round (Q. Yeates-Burghart and C. O'Brien, unpublished results). Future studies might consider whether a lower threshold expression of TSH β 1 is required to initiate the cascade of events leading sexual maturation in *G. aculeatus*.

Gonadotropin releasing hormone

The cascade of response to gonad-stimulating long days in *G. aculeatus* continues with a change in the level of gonadotropin releasing hormone (GnRH), but with a decrease (Fig. 4.6) rather than an increase in expression, as is seen in photoperiodic tetrapods (Yasuo & Yoshimura, 2009). We propose three potential explanations.

First, regulation of LH may be independent of GnRH in stickleback. However, GnRH stimulates LH secretion throughout vertebrates, including photoperiodic fishes (Borg, 2010). In stickleback, the pars distalis has extensive innervation from GnRH neurons (Andersson *et al.*, 1995). As the pars distalis is the site of gonadotropin production and secretion in the vertebrate pituitary (Kah & Dufour, 2010), this innervation supports the concept of a direct control of gonadotropins by GnRH.

Second, GnRH3 may not be the actual hypophysiotropic paralog of GnRH. Although the distribution of GnRH2 and GnRH3 expression is similar to that in other teleost fishes that have lost GnRH1 (Anderson *et al.*, 1995; Okubo & Nagahama, 2008), GnRH3 and not GnRH2 is the hypophysiotropic form in other species of fish that lack GnRH1 (Chen & Fernald, 2008; Cerdá-Reverter & Canosa, 2009). Future research should consider the possibility that GnRH2 may be the hypophysiotropic form in *G*. *aculeatus* as well as other teleosts.

Third, the hypophysiotropic function of GnRH1 and, in the fish species where it is absent, GnRH3, is well documented, but additional functions of either paralog are much less understood (Chen & Fernald, 2008). In species where GnRH3 has assumed the hypophysiotropic function, we would expect it to retain its other functions as well. If GnRH3 expression is inhibited by long days in areas of the brain related to these other functions, but simultaneously stimulated in areas related to its hypophysiotropic function, the net expression of GnRH3 in the brain and pituitary combined could still decrease during long days. Sexual maturation in the grey mullet is regulated by photoperiod (Kuo *et al.*, 1974) and it has retained GnRH1 as the hypophysiotropic GnRH (Nocillado *et al.*, 2007). GnRH3 expression decreases in the brain of the grey mullet during sexual

maturation (Nocillado *et al.*, 2007), presumably in the context of its other, nonhypophysiotropic functions. Future research should probe the other functions of GnRH paralogs, unrelated to gonadal maturation, especially in photoperiodic fish.

Luteinizing hormone

LH β is the third hormone to be expressed in the sequence of events leading downstream from gonadal stimulating long days (Table 4.1 and Fig. 4.6). LH β is a gonadotropin that stimulates the sex hormones required to initiate sexual maturation in vertebrates. In the Japanese quail, a single long day stimulates LH β release (Nakao *et al.*, 2008). A single long day also affects phenotypic indicators of sexual maturation in at the photoperiodic Siberian hamster (Finley *et al.*, 1995). In the threespine stickleback, LH β expression rises above baseline after 5-10 long days but, given the effects of a single long day on quail and hamsters, the later expression in LH β does not necessarily mean that 5-10 long days are necessary for LH β expression or to commit stickleback to sexual maturation. Future research should determine the number of long days required to activate the entire TSH β to LH β cascade in stickleback and whether, once increased above baseline, expression of LH β is sufficient to commit stickleback to seasonal reproductive maturation.

Conclusions

Our results strongly support a direct role for the TH pathway in the photoperiodic initiation of sexual maturation in teleosts, supporting the functional conservation of the TH pathway among photoperiodic vertebrates. Our use of lab-raised populations and

controlled experimental conditions allowed us to eliminate the potential influence of other environmental or historical factors that have limited inference in previous studies of the physiological basis of photoperiodic response in teleost fish. Although the photoperiodic responses of the populations in this study are phenotypically indistinguishable (O'Brien *et al.*, in prep), their physiological responses demonstrate the benefits of replicating studies across multiple populations. First, the differences in early TSH β 1 response between the southern and northern populations may reflect differences in seasonal reproductive patterns. Second, the initial decrease in GnRH3 was unexpected, but is robust because this decrease was consistent among these populations. Future work motivated by these GnRH3 results will illuminate functional variation in a highly conserved hormone family (Chen & Fernald, 2008). Finally, the gradual LHB increases in all populations suggests differences between fish and birds in the timing of sexual maturation in response to photoperiod. To our knowledge, the ecological significance of the early LH release in birds has not been addressed. The results herein are motivation and a basis for such studies.

Taken together, our results further establish the threespine stickleback as a vertebrate model of photoperiodic response (Borg *et al.*, 2004) and form a foundation for future investigations into the hormonal basis of vertebrate photoperiodic response in varied seasonal environments.

Bridge

Chapter IV examines the physiological basis of early photoperiodic response using the threespine stickleback as a model teleost fish. We show that the thyroid

hormone pathway initiates sexual maturation, which strongly suggests that the hormonal and anatomical basis of photoperiodic response is conserved among vertebrates. In Chapter V, I conclude by summarizing the data presented in this dissertation, their significance, and suggest ways in which the threespine stickleback can contribute as a model of vertebrate photoperiodism

CHAPTER V

CONCLUSION

The presence of biological clocks among nearly all forms of life underscores the importance of correct time perception for dealing with periodic environmental variation. The circadian clock controls the timing of daily organismal activities. The timing of seasonal actions in polar and temperate organisms is controlled by the photoperiodic timer, which is set by photoperiod (length of day). The highly reliable annual cycle of photoperiod allows animals to anticipate and properly prepare for future environmental conditions so that these seasonal actions occur at the optimal time of year. Despite its ecological and evolutionary importance, very little is known about the genetic basis of photoperiodic interpretation in natural vertebrate populations. My dissertation research was motivated by a desire to address this question. It does so demonstrating the proper methodology for studying photoperiodism and establishing the threespine stickleback as model of vertebrate photoperiodism.

How should a complex trait like vertebrate photoperiodism be studied?

Proper study of the genetic basis of photoperiodism requires a clear understanding of the assumptions underlying interpretation of results and the necessary attributes of a model organism whose study will produce solid progress. Our work herein addresses the effects of unreasonable assumptions on the state of the field and establishes the threespine stickleback as a model that can overcome limitations that have hindered progress.

The assumption that the circadian clock forms the basis of the photoperiodic timer in animals is widespread, but has not been demonstrated in natural populations (Bradshaw & Holzapfel, 2007). In O'Brien *et al.* (2011; Chapter II), we examine the correlation between a proxy for the circadian clock (allelic variation in a circadian clock gene) and the timing of migration in several salmonid species as a proxy for photoperiodic time measurement. This correlation has been interpreted as support for a causal connection between the circadian clock and photoperiodic time measurement (O'Malley *et al.*, 2010).

This interpretation rests on the key assumptions that the circadian clock is functionally integrated with the photoperiodic timer and that genetic variation in a circadian clock gene must affect the photoperiodic timer through its role in the circadian clock. The role of the circadian clock in seasonal timing is debated (Hazlerigg & Loudon, 2008; Goto *et al.*, 2010; Bradshaw & Holzapfel, 2010a, 2010b; Saunders, 2010; Koštál, 2011; Schiesari *et al.*, 2011), so causal relationships among all these elements must be established to support the hypothesis. We described a simple, but underemployed method that can be employed before this work is undertaken to determine if the association between the proxy of the circadian clock and migratory timing remains once their correlation due to their common covariance with latitude is removed. We found that it does not, which means their putative relationship was due to autocorrelation caused by a

common association with latitude. Thus, further investigation of a causal relationship is inappropriate.

The result of a lack of covariation between latitude and allelic variation in the candidate gene demonstrates both the value of applying this test as an initial step before examination of a correlation across an ecogeographic gradient and that caution must be employed when inferring causality between the circadian clock and photoperiodic time measurement, especially when employing proxies for each.

In Chapter III, we addressed the hypothesized association between a proxy of the circadian clock, the eponymous gene *clock*, and the response of traits mediated by photoperiod. First, we demonstrated that a stimulatory photoperiod elicits sexual maturation in male and female threespine stickleback. We developed a method and the equipment necessary to phenotype photoperiodic response in conditions that controlled for environmental variables that may affect output from the photoperiodic timer, which include temperature, nutrient availability, water quality, hormonal cues, and visual cues.

The ability to directly assess output of the photoperiodic timer means that our interpretation of the results is unhindered by the potentially confounding effects of other environmental signals. Isolating the effects of individual variables is important so as to not draw incorrect causal connections. This point is well demonstrated by considering salmonid migratory timing, which is mediated by the photoperiodic timer, but is also affected by local temperature (Crozier *et al.*, 2008) and perhaps other environmental signals. O'Malley *et al.* (2010) assumed that observed variation in migratory timing is strictly due to variation in photoperiodic response. This is not a reasonable assumption as the effects of these other environmental signals are not measured or controlled.

This assumption underlies the conclusions of O'Malley et al. (2010) that a correlation between variation in a circadian clock gene and migratory timing supports a role for the circadian clock in the photoperiodic timer. We tested this conclusion by examining allelic variation in the circadian clock gene among threespine stickleback populations that had been phenotyped for photoperiodic response in controlled conditions. By examining a potential correlation strictly with photoperiodic response and not phenological variation that may be affected by other environmental signals we were able to conduct a much stricter test of the hypothesized relationship. Although there is a high degree of variation in the circadian clock gene, it is not associated with populations or latitude. We conclude that there is not a causal connection between this component of the circadian clock and the photoperiodic timer. This conclusion complements that of Chapter II, by empirically demonstrating a lack of association between allelic variation and latitude in another photoperiodic teleost fish. The results of Chapters II and III show that interpretation of genetic variation along a geographic gradient should be done cautiously. In particular, suggesting that the previously observed allelic variation in a circadian clock gene is related to variation in seasonal activities mediated by photoperiodic response is imprudent at best.

Is the hormonal basis of photoperiodic response conserved among vertebrates?

Many temperate and polar vertebrates use photoperiod to mediate sexual maturation (Bromage *et al.*, 2001; Bradshaw & Holzapfel, 2007). The thyroid hormone (TH) pathway initiates this process in photoperiodic mammals and birds (Yasuo & Yoshimura, 2009), but it is unknown if the pathway's function and stimulation by

photoperiod is conserved in teleost fishes. In Chapter IV, we measure the response of key hormones in the threespine stickleback TH pathway in response to a stimulatory photoperiod. The data show that these hormones generally have the same response and neuroanatomical location in threespine stickleback as they do in mammals and birds (Chapter IV). This strongly supports conservation of the TH pathway's role among mammals, birds, fishes and perhaps all of vertebrates. The advantages of examining photoperiodic response in controlled conditions are also demonstrated, as the changes in hormone levels can be attributed solely to photoperiod, thus making the results directly comparable to those from mammal and bird models.

sWhat makes a good model of vertebrate photoperiodism?

The second way in which this dissertation addresses the main motivation of understanding the physiological and genetic basis of photoperiodism is by establishing the threespine stickleback as a vertebrate model of photoperiodic response. In general, there are two types of vertebrate models used in studies of photoperiodism. The first type comprises species that are of ecological or economic interest, but are difficult to raise, maintain, and/or manipulate in controlled conditions. Studies of these species, such as salmonids, passerines, and wild rodents, often rely upon assumed proxies of photoperiodic time measurement, such as migratory timing, metabolic dormancy, reproductive maturation or reproductive quiescence. These phenotypes may be mediated by photoperiod, but the impracticality of studying them in strictly controlled conditions means that the effects of other environmental signals upon their timing, such as temperature, nutrient availability and presence of con-specifics, are unmeasured. We are therefore unable to attribute any observed variation in seasonal timing strictly to variation in photoperiodic response. As a result, our ability to make inferences regarding the contribution of photoperiodism to life history modes and adaptation to local environments in vertebrates is compromised.

The second category of vertebrate models include organisms that are practical to manipulate in controlled laboratory conditions and are amenable to the techniques necessary to understand the processes underlying phenotypic responses to photoperiod, such as hamsters, Soay sheep, and Japanese quail. Studies using such models have greatly advanced our understanding of the transcriptional, hormonal and anatomical basis of photoperiodism. Major findings include the earliest indicators of the transcriptional and hormonal responses to long days and the conserved nature of these responses in mammals and birds (Yasuo & Yoshimura, 2009). However, studies using this type of model have not considered or are unable to consider how intraspecific comparisons of different life histories or latitudes may inform our understanding of vertebrate photoperiodism. In addition, these results may be affected by inadvertent evolution of the study lines caused by their maintenance in small populations over many generations in laboratory conditions, which may result in evolution due to inbreeding and/or adaptation to an unnatural environment. The effects of reliance on single populations or laboratory maintained lines on our understanding of the transcriptional and physiological basis of photoperiodic response is unknown. At the least, it limits our ability to address these aspects of photoperiodism in natural vertebrate populations and, in the future, to understand the underlying basis of among population variation in vertebrate photoperiodic response.

The threespine stickleback has the advantages of both these types of models. Like the first type, it inhabits a wide latitudinal (i.e. climatic) range and has several distinct life history strategies (Bell & Foster, 1994). Like the second type, its photoperiodic response can be phenotyped in laboratory conditions that control for the potential effects of other environmental signals (Yeates-Burghart *et al.*, 2009; Chapter III) and it is amenable to the techniques necessary to understand the basis of phenotypic responses to photoperiod (Cresko *et al.*, 2007; Chapter IV). Chapters III and IV demonstrate how these advantages are employed to advance our understanding of vertebrate photoperiodism.

Chapter III measures the photoperiodic response of multiple populations from Alaska and Oregon. The severity of seasonality increases with latitude, so the consequences for the mistiming of seasonal behavior are also expected to increase. This is hypothesized to result in a greater reliance on photoperiod as a predictable indicator of seasonal change at higher latitudes (Bradshaw & Holzapfel, 2007). Although there are multiple examples of variation in invertebrate photoperiodic response and proxies of vertebrate photoperiodic response over such a latitudinal range (reviewed in Danilevskii, 1965; Bradshaw & Holzapfel, 2007), there is no difference among populations or between the two latitudes in photoperiodic response. We suggest that the lack of interlatitudinal difference may result from differences in how the populations respond to temperature in the wild.

These data are the first study of photoperiodic response across multiple populations of a vertebrate in controlled conditions. They demonstrate that outbred lines created from multiple populations are practical to raise and maintain in a common environment and that the photoperiodic response of these lines can be measured in

conditions that control for the effects of other environmental signals. As such, they are the initial steps in establishing the threespine stickleback as a model of vertebrate photoperiodism.

How will the threespine stickleback inform our understanding of photoperiodism in relation to life history and local environment?

Threespine stickleback occur from northern Alaska (Bell & Foster, 2004) to as far south as the Baja peninsula on the west coast of North America (Sánchez-Gonzáles *et al.*, 2002). Future work regarding the relationship between the severity of seasonal change and the reliance upon a reliable, predictive cue of it should take advantage of this large latitudinal range. We expect that more southern populations than Oregon and Alaska will rely less on photoperiod and more on proximate environmental signals, such as nutrient availability and water temperature, as cues for the initiation of sexual maturation. This work would be the first to address latitudinal trends in vertebrate photoperiodism.

The threespine stickleback is also a valuable model for understanding the effects of life history on photoperiodism. Very little is known about how photoperiodism constrains, facilitates, or otherwise influences adaptation to an environment. There are several examples in vertebrates of the rapid evolution of seasonal timing after introduction to a new environment or during environmental change (Quinn & Adams, 1996; Quinn *et al.*, 2000; Réale *et al.*, 2003; Bearhop *et al.*, 2005; Møller, 2007). These traits are mediated by photoperiodism, but the relative contributions of photoperiodism, response to other environmental signals, or phenotypic plasticity to the observed changes have not been determined.

Many of the divergent life history forms produced in the threespine stickleback adaptive radiation have been used as behavioral, ecological, and, more recently, genomic models (Bell & Foster, 1994; Kingsley et al., Cresko et al., 2007). This background knowledge would aid the selection of populations for the phenotyping of photoperiod response and inform interpretation of results regarding the effects of life history on photoperiodism. For instance, adaptive morphological differences are maintained over small spatial scales between populations of lake and stream stickleback populations in British Columbia (Hendry et al., 2002; Berner et al., 2008). Lake stickleback sexually mature several weeks later than stream stickleback (A. Hendry, pers. comm.). This difference is genetic, as it maintained in laboratory conditions (A. Hendry, pers. comm.), but it is unclear if it is a result of variation in photoperiodic response or temperature response. This can be tested using the straightforward methods we demonstrated in Chapter III. If there is variation in photoperiodic response between these lake and stream populations, it suggests that such variation may quickly evolve as an isolating mechanism for the maintenance of local adaptation in the presence of maladaptive gene flow.

This example demonstrates the potential of the stickleback model to inform our understanding of how photoperiodism evolves during adaptation to an environment, its interactions with life history, and its potential as a mechanism to facilitate adaptation. Such studies will provide a foundation for understanding the genetic and physiological bases of photoperiodism.

How will the threespine stickleback inform our understanding of the genetic basis of photoperiodism?

Most laboratory studies on vertebrate photoperiodism have focused on the processes underlying photoperiodic response, at the expense of addressing the genetic basis of photoperiodic interpretation. Many studies that have addressed this question have done so using the candidate gene approach, often under the assumption that the circadian clock forms the basis of the photoperiodic timer. As is discussed in Chapter I and demonstrated in Chapters II and III, the evidence for this is equivocal at best. Actual progress will come from the forward genetic approach, which is unconstrained by selecting a priori candidates for examination.

Forward genetics is initially more difficult than the candidate gene approach, as it requires understanding a complex trait such as photoperiodism in a way that makes it amenable to genetic dissection. Recent work in mammals, birds (Nakao *et al.*, 2008; Dardente *et al.*, 2010; Masumoto *et al.*, 2010), and now a teleost fish (Chapter IV) that defined the earliest transcriptional responses to a stimulatory photoperiod has made this possible. This early response can be used as a time point around which to conduct tissue specific sampling of transcribed genes. Those that differ in expression between organisms in stimulatory and non-stimulatory photoperiods will be involved in either interpretation of photoperiod or the initial response.

These genes will be the basis of two types of follow-up studies. The first is to determine their function: where and when they are expressed, what other genes they interact with, and the phenotypic results of interfering with their expression. This work is

necessary to establish evidence of their causal involvement in photoperiodic interpretation beyond the initial correlation of differential expression with photoperiodic response.

The second type of study proceeds from the first: if a subset of these genes is functionally involved in photoperiodic interpretation, it is necessary to determine if they vary in natural populations and if this variation is associated with phenotypic variation in photoperiodism. A gene may be functionally involved in a trait without being involved in evolution of that trait. This may be due to constraints on its evolution resulting from pleiotropy, the function of its protein product, or a lack of genetic variation within it that can respond to selection. Thus, a survey of these genes in natural populations that vary in photoperiodic response is necessary to understand which are actually involved in the evolution of photoperiodism. Such work will rely on establishing phenotypic variation in photoperiodism among populations, as described in the previous section.

Conclusion: how studies of vertebrate photoperiodism will inform our understanding of evolution

Determining which genes underlie the apparent ability of the photoperiodic timer to rapidly evolve in response to a changing environment (Quinn & Adams, 1996; Quinn *et al.*, 2000; Réale *et al.*, 2003; Bearhop *et al.*, 2005; Møller, 2007) will allow us to make a connection between genotype and phenotype in a complex trait that is essential for organismal fitness in the wild. Connecting genetic variation to phenotypic variation is a central goal of evolutionary genetics (Lewontin, 1974). The modern field of evolution of development was founded in large part because of the realization that this requires understanding the physiological and cellular processes that connect these two levels of biological organization. In this context, the study of vertebrate photoperiodism will provide valuable insight into how a trait whose downstream physiological response can be conserved across hundreds of millions of years of evolution (Yasuo & Yoshimura, 2009; Chapter IV) can also rapidly evolve in response to a changing environment. Although it is a complex trait, whose study is made more difficult by the timescale on which its output occurs, this conservation suggests that it may be a more tractable phenotype for genetic dissection than it first appears and that results from an appropriate model organism may be generalizable across vertebrates.

The importance of proper seasonal timing for organismal fitness and its prevalence in polar and temperate organisms validates our work to understand photoperiodism on multiple levels of biological organization. The threespine stickleback is an excellent model for such studies because of its wide latitudinal range, varied life history and our ability to raise and manipulate outbred populations in controlled conditions. The work presented herein establishes it as a model of vertebrate photoperiodism and forms the foundation for future studies. The results will further our understanding of the timing of organismal processes in a seasonal environment and the connection between genotype and phenotype in a widespread and ecologically relevant trait.

APPENDIX A

SUPPLEMENTARY INFORMATION: CHAPTER II

These supplementary Excel files can be downloaded separately from the dissertation.

Appendix S2.1, Regressions and calculations used to generate Fig. 2 in the main text. File title: "MEC_5133_sm_AppendixS1.xls."

Appendix S2.2. Data, regression and ANOVAs for average PolyQ length, run time, and latitude for Fig. 4 in the main text. File title: "MEC_5133_sm_AppendixS2.xls."
APPENDIX B

SUPPLEMENTARY INFORMATION: CHAPTER III



Supplementary Fig. 3.1. Syntenic analysis of the *G. aculeatus clock* gene. Gray dots indicate genes found on *G. aculeatus* linkage group IX. Red crosses indicate orthologs in the *H. sapiens* genome. *H. sapiens* (ENSG00000134852) and *G. aculeatus* (ENSGACG00000015939) *clock* homologs are indicated by blue circles.

APPENDIX C

SUPPLEMENTARY INFORMATION: CHAPTER IV

		Photoperiod Treatment				
_		Day 0	Day 1	Day 2	Day 5	Day 10
Population	North 1	10	12	14	10	10
	North 2	10	10	10	10	8
	South	8	10	8	8	8

Supplementary Table 4.1. Sample sizes per treatment for quantitative PCR experiments. All fish were adult males.

Supplementary Fig. 4.1. (next page). Phylogenetic reconstructions. For all trees, the best model was JTT (Jones et al., 1992) with a gamma-distributed set of evolutionary rates (Zang et al., 1996). Although the preprohormones of the target orthologs are short (90 – 150 amino acids), and thus have poor phylogenetic signal, the G. aculeatus orthologs measured in this study are placed in the expected clades with high support. SH-like support scales the likelihood ratio test between 0 - 1. Nodes with values greater than .7 are considered highly supported. Branch lengths are proportional within phylogenies, except where hash marks indicate truncation. Clades with poor internal branch support are collapsed. Species within collapsed clades have been alphabetized by genus name. SH-like support values are listed at their respected nodes. (A) TSHB. Homo sapiens LHB was used to root the TSH^β phylogeny. There is a strong separation between tetrapod and teleost TSHB orthologs, although incorrect placement of *B. japonicus* makes the tetrapod TSH^β paraphyletic. This is a result of the poor phylogenetic signal, as indicated by the weak node support. The threespine stickleback TSHB1 ortholog measured in this study is within the expected clade. (B) GnRH. The *Petromyzon marinus* GnRH paralogs were used to root this phylogeny. There is strong support for monophyly of the three GnRH forms. As expected, GnRH3 is nested within the GnRH3 clade, as GnRH3 is unique to teleosts (Chen & Fernald, 2008). As expected, the G. aculeatus GnRH3 ortholog measured in this study is within this clade. (C) LHB. Homo sapiens TSHB was used to root the gonadotropin phylogeny. There is strong support for distinct FSHB and LHB clades. Within these, there is strong support for tetrapod and teleost clades, although the incorrect placement of X. laevis LHB and N. forsteri LHB makes the tetrapod LHB clade paraphyletic and lowers the node support for the division between it and the teleost LHB clade. As expected, the G. aculeatus LHB ortholog measured in this present study is within the teleost LHB clade.



.811

H. sapiens TSHβ

0.792

 $\begin{array}{c} \text{Japonica LH}_{\beta} \\ \text{Japonica LH}_{\beta} \\ \text{Lmsculus LH}_{\beta} \\ \text{norvegicus LH}_{\beta} \\ \end{array} \\ \begin{array}{c} \text{LH}_{\beta} \\ \text{LH}_{\beta} \\ \end{array}$

REFERENCES CITED

- Álvarez-Viejo, M., Cernuda-Cernuda, R., Álvarez-López, C., & García-Fernández, J. M. (2004). Identification of extraretinal photoreceptors in the teleost *Phoxinus phoxinus*. *Histology and Pathology* **19**, 487-494.
- Akaike, H. (1973). Information theory and an extension of the maximum likelihood principle. In: *Proceedings of the Second International Symposium on Information Theory*, (Petrov, B. H., Caski, F. eds), pp. 267-281. Budapest: Akademiai Kiado.
- Amano, M., Hyodo, S., Kitamura, S., Ikuta, K., Suzuki, Y., Urano, A., & Aida, K. (1995). Short photoperiod accelerates preoptic and ventral telencephalic salmon GnRH synthesis and precocious maturation in underyearling male masu salmon. *General and Comparative Endocrinology* 99, 22-27.
- Amano, M., Ikuta, K., Kitamura, S., & Aida, K. (2001). Effects of photoperiod on pituitary gonadotropin levels in masu salmon. *Journal of Experimental Zoology* 289, 449-455.
- Amano, M., Ikuta, K., Kitamura, S., & Aida, K. (1999). Effects of photoperiod on salmon GnRH mRNA levels in brain of castrated underyearling precocious male masu salmon. *General and Comparative Endocrinology* **115**, 70-75.
- Andersson, E., Bogerd, J., Borg, B., Sharp, P. J., Sherwood, N. M., & Goos, H. J. T. (1995). Characterization and localization of gonadotropin-releasing hormone in the brain and pituitary of the three-spined stickleback, *Gasterosteus aculeatus*. *Cell and Tissue Research* 279, 485-493.
- Andersson, E., Borg, B., & Goos, H. J. T. (1992). Temperature, but not photoperiod, influences gonadotropin-releasing hormone binding in the pituitary of the threespined stickleback, *Gasterosteus aculeatus*. *General and Comparative Endocrinology* 88, 111 - 116.
- Angilletta, M. J. (2001). Variation in metabolic rate between populations of a geographically widespread lizard. *Physiological and Biochemical Zoology* 74, 11-21.
- Anisimova, M. & Gascuel, O. (2006). Approximate likelihood-ratio test for branches: A fast, accurate, and powerful alternative. *Systematic Biology* **55**, 539-552.
- Avivi, A., Albrecht, U., Oster, H., Joel, A., Beiles, A., & Nevo, E. (2001). Biological clock in total darkness: The Clock/MOP3 circadian system of the blind subterranean mole rat. *Proceedings of the National Academy of Sciences USA* 98, 13751-13756.

- Baggerman, B. (1985). The roles of daily and annual biological rhythms in the photoperiodic regulation of the breeding season in the stickleback *Gasterosteus aculeatus* L. *Behaviour* **93**, 1-7.
- Baggerman, B. (1972). Photoperiodic responses in the stickleback and their control by a daily rhythm of photosensitivity. *General and Comparative Endocrinology* **3**, 466-476.
- Baggs, J. E., Price, T. S., DiTacchio, L., Panda, S., FitzGerald, G. A., & Hogenesch, J. B. (2009). Network Features of the Mammalian Circadian Clock. *PLoS Biology* 7, e1000052.
- Bearhop, S., Fiedler, W., Furness, R. W., Votier, S. C., Waldron, S., Newton, J., Bowen, G. J., Berthold, P., & Farnsworth, K. (2005). Assortative mating as a mechanism for rapid evolution of a migratory divide. *Science* **310**, 502-504.
- Bedecarrats, G., Shimizu, M., & Guemene, D. (2006). Gonadotropin releasing hormones and their receptors in avian species. *Journal of Poultry Science* **43**, 199 - 214.
- Bell, M. A. & Foster, S. A. (1994). The Evolutionary Biology of the Threespine Stickleback. London: Oxford University Press.
- Bentley, G. E., Moore, I. T., Sower, S. A., & Wingfield, J. C. (2004). Evidence for a novel gonadotropin-releasing hormone in hypothalamic and forebrain areas in songbirds. *Brain, Behavior and Evolution* 63, 34-46.
- Berner, D., Adams, D. C., Grandchamp, A. C., & Hendry, A. P. (2008). Natural selection drives patterns of lake--stream divergence in stickleback foraging morphology. *Journal of Evolutionary Biology* 21, 1653-1665.
- Biswas, A., Kundu, S., Roy, S., De, J., Pramanik, M., & Ray, A. K. (2006). Thyroid hormone profile during annual reproductive cycle of diploid and triploid catfish, *Heteropneustes fossilis* (Bloch). *General and Comparative Endocrinology* 147, 126-132.
- Blackman, B. K. (2010). Connecting genetic variation to phenotypic clines. *Molecular Ecology* 19, 621-623.
- Borg, B. (2010). Photoperiodism in Fishes. In: *Photoperiodism: the Biological Calender*, (Nelson, R. J, Denlinger, D. L. & Somers, D. E., eds), pp. 371-398. New York: Oxford University Press.
- Borg, B. (1982). Seasonal effects of photoperiod and temperature on spermatogenesis and male secondary sexual characters in the three-spined stickleback, *Gasterosteus* aculeatus L. Canadian Journal of Zoology 60, 3377-3386.

- Borg, B. (1987). Stimulation of reproductive behaviour by aromatizable and nonaromatizable androgens in the male three-spined stickleback, *Gasterosteus aculeatus* L. In *Proceedings of the 5th Congress of European Ichthyologists* (Kullander, S. A. & Fernhom, B., eds), pp. 269-271, Stockholm: Department of Vertebrate Zoology, Swedish Museum of Natural History.
- Borg, B., Bornestaf, C., Hellqvist, A., Schmitz, M., & Mayer, I. (2004). Mechanisms in the photoperiodic control of reproduction in the stickleback. *Behaviour* 131, 1521-1530.
- Borg, B. & Van Veen, T. (1982). Seasonal effects of photoperiod and temperature on the ovary of the three-spined stickleback, *Gasterosteus aculeatus* L. *Canadian Journal of Zoology* **60**, 3387 - 3393.
- Bornestaf, C. & Borg, B. (2000). Endogenous breeding cycles in male threespine sticklebacks, *Gasterosteus aculeatus*. *Behaviour* **137**, 921-932.
- Bornestaf, C., Mayer, I., & Borg, B. (2001). Melatonin and maturation pace in female three-spined stickleback, *Gasterosteus aculeatus*. *General and Comparative Endocrinology* **122**, 341-348.
- Bradshaw, W. E. & Holzapfel, C. M. (2007). Evolution of animal photoperiodism. Annual Review of Ecology, Evolution, and Systematics **38**, 1-25.
- Bradshaw, W. E. & Holzapfel, C. M. (2001). Genetic shift in photoperiodic response correlated with global warming. *Proceedings of the National Academy of Sciences* USA 98, 14509-14511.
- Bradshaw, W. E. & Holzapfel, C. M. (2010a). What season is it anyway? Circadian tracking vs. photoperiodic anticipation in insects. *Journal of Biological Rhythms* 25, 155-165.
- Bradshaw, W. E. & Holzapfel, C. M. (2007). Tantalizing timeless. *Science* **316**, 1851-1852.
- Bradshaw, W. E. & Holzapfel, C. M. (2010b). Circadian clock genes, ovarian development and diapause. *BMC Biology* **8**, 115.
- Bradshaw, W. E. & Holzapfel, C. M. (2010). Light, time, and the physiology of biotic response to rapid climate change in animals. *Annual Review of Physiology* 72, 147-166.
- Bradshaw, W. E., Zani, P. A., & Holzapfel, C. M. (2004). Adaptation to temperate climates. *Evolution* **58**, 1748-1762.
- Bromage, N., Porter, M., & Randall, C. (2001). The environmental regulation of maturation in farmed finfish with special reference to the role of photoperiod and melatonin. *Aquaculture* 197, 63-98.
- Bünning, E. (1936). Die endogene Tagesrhythmik als Grundlage der photoperiodischen Reaktion. *Berichte der Deutschen Botanischen Gesellschaft* **54**, 590-607.

- Caicedo, A. L., Stinchcombe, J. R., Olsen, K. M., Schmitt, J., & Purugganan, M. D. (2004). Epistatic interaction between *Arabidopsis* FRI and FLC flowering time genes generates a latitudinal cline in a life history trait. *Proceedings of the National Academy of Sciences USA* **101**, 15670-15675.
- Carlson, L. L., Zimmermann, A., & Lynch, G. R. (1989). Geographic differences for delay of sexual maturation in Peromyscus leucopus: effects of photoperiod, pinealectomy, and melatonin. *Biology of Reproduction* 41, 1004-1013.
- Cassone, V. (1998). Melatonin's role in vertebrate circadian rhythms. *Chronobiology International* **15**, 457-473.
- Catchen, J. M., Braasch, I., & Postlethwait, J. H. (2011). Conserved synteny and the zebrafish genome. In: *The Zebrafish: Genetics, Genomics, and Informatics* (Dietrich, W. H., Westerfield, M. Zon, L. I., eds.), pp. 256-268. Amsterdam: Elsevier Science.
- Catchen, J. M., Conery, J. S., & Postlethwait, J. H. (2009). Automated identification of conserved synteny after whole-genome duplication. *Genome Research* 19, 1497 -1505.
- Cerdá-Reverter, J. M. & Canosa, L. F. (2009). Anatomical neuroendocrinology. In: *Fish Physiology*, (Bernier N. J., Van Der Kraak, P. & Anthony C. J., eds), pp. 4-55. Amsterdam: Elsevier Science.
- Chaurasia, S. S., Rollag, M. D., Jiang, G., Hayes, W. P., Haque, R., Natesan, A., Zatz, M., Tosini, G., Liu, C., Korf, H. W., Iuvone, P. M., & Provencio, I. (2005).
 Molecular cloning, localization and circadian expression of chicken melanopsin (Opn4): differential regulation of expression in pineal and retinal cell types. *Journal of Neurochemistry* 92, 158 170.
- Chen, C. C. & Fernald, R. D. (2008). GnRH and GnRH receptors: distribution, function and evolution. *Journal of Fish Biology* **73**, 1099 1120.
- Childs, G. V. (2006). Gonadotropes and lactotropes. In: *Physiology of Reproduction*, (Neill, J. & Knobil E., eds), pp. 1483-1579. New York: Elsevier Press.
- Choi, S., Lee, C. H., Park, W., Kim, D. J., & Sohn, Y. C. (2010). Effects of shortened photoperiod on gonadotropin-releasing hormone, gonadotropin, and vitellogenin gene expression associated with ovarian maturation in rainbow trout. *Zoological Science* 27, 24-32.
- Clarke, W., Withler, R., & Shelbourn, J. (1994). Inheritance of smolting phenotypes in backcrosses of hybrid stream-type × ocean-type chinook salmon (*Oncorhynchus tshawytscha*). *Estuaries and Coasts* **17**, 13-25.
- Cresko, W. A., Amores, A., Wilson, C., Murphy, J., Currey, M., Phillips, P., Bell, M. A., Kimmel, C. B., & Postlethwait, J. H. (2004). Parallel genetic basis for repeated evolution of armor loss in Alaskan threespine stickleback populations. *Proceedings of the National Academy of Sciences USA* **101**, 6050-6055.

- Cresko, W. A., McGuigan, K., Phillips, P., & Postlethwait, J. H. (2007). Studies of threespine stickleback developmental evolution: progress and promise. *Genetica* **129**, 105-126.
- Crivelli, A. & Britton, R. (1987). Life history adaptations of *Gasterosteus aculeatus* in a Mediterranean wetland. *Environmental Biology of Fishes* **18**, 109 125.
- Crozier, L. G., Hendry, A. P., Lawson, P. W., Quinn, T. P., Mantua, N. J., Battin, J., Shaw, R. G., & Huey, R. B. (2008). Potential responses to climate change in organisms with complex life histories: evolution and plasticity in Pacific salmon. *Evolutionary Applications* 1, 252-270.
- Cruciani, F., Trombetta, B., Labuda, D., Modiano, D., Torroni, A., Costa, R., & Scozzari, R. (2008). Genetic diversity patterns at the human clock gene *period 2* are suggestive of population-specific positive selection. *European Journal of Human Genetics* 16, 1526-1534.
- Curwen, V., Eyras, E., Andrews, T. D., Clarke, L., Mongin, E., Searle, S. M. J., & Clamp, M. (2004). The Ensembl automatic gene annotation system. *Genome Research* 14, 942-950.
- Cyr, D. G., Bromage, N. R., Duston, J., & Eales, J. G. (1988). Seasonal patterns in serum levels of thyroid hormones and sex steroids in relation to photoperiod-induced changes in spawning time in rainbow trout, *Salmo gairdneri*. *General and Comparative Endocrinology* 69, 217 - 225.
- Cyr, D. G. & Eales, J. G. (1996). Interrelationships between thyroidal and reproductive endocrine systems in fish. *Reviews in Fish Biology and Fisheries* **6**, 165 200.
- Dalziel, A. C., Rogers, S. M., & Schlute, P. M. (2009). Linking genotypes to phenotypes and fitness: how mechanistic biology can inform molecular ecology. *Molecular Ecology* 18, 4997-5017.
- Danilevskii, A. S. (1965). *Photoperiodism and Seasonal Development in Insects*. Edinburgh: Oliver & Boyd.
- Dardente, H., Klosen, P., Pévet, P., & Masson-Pévet, M. (2003). MT1 Melatonin receptor mRNA expressing cells in the pars tuberalis of the european hamster: effect of photoperiod. *Journal of Neuroendocrinology* 15, 778-786.
- Dardente, H., Wyse, C. A., Birnie, M. J., Dupré, S. M., Loudon, A. S. I., Lincoln, G. A., & Hazlerigg, D. G. (2010). A molecular switch for photoperiod responsiveness in mammals. *Current Biology* 20, 2193-2198.
- Darlington, T. K., Wager-Smith, K., Ceriani, M. F., Staknis, D., Gekakis, N., Steeves, T. D. L., Weitz, C. J., Takahashi, J. S., & Kay, S. A. (1998). Closing the circadian loop: CLOCK-induced transcription of its own inhibitors *per* and *tim. Science* 280, 1599-1603.
- Davies, B., Bromage, N., & Swanson, P. (1999). The brain-pituitary-gonadal axis of female rainbow trout Oncorhynchus mykiss: effects of photoperiod manipulation. General and Comparative Endocrinology 115, 155-166.
- Dawson, A. (2002). Photoperiodic control of the annual cycle in birds and comparison with mammals. *Ardea* **90**, 355-367.

- Dawson, A. & Sharp, P. J. (2007). Photorefractoriness in birds--photoperiodic and nonphotoperiodic control. *General and Comparative Endocrinology* 153, 378-384.
- Dubois, E. A., Zandbergen, M. A., Peute, J., & Goos, H. J. (2002). Evolutionary development of three gonadotropin-releasing hormone (GnRH) systems in vertebrates. *Brain Research Bulletin* 57, 413-418.
- Dunlap, J. C., Loros, J. J., & Decoursey, P. J. (2004). Chronobiology: Biological Timekeeping. Sunderland: Sinauer Associates.
- Edgar, R. C. (2004). MUSCLE: multiple sequence alignment with high accuracy and high throughput. *Nucleic Acids Research* **32**, 1792-1797
- Emerson, K. J., Bradshaw, W. E., & Holzapfel, C. M. (2008). Concordance of the circadian clock with the environment is necessary to maximize fitness in natural populations. *Evolution* 62, 979-983.
- Emerson, K. J., Bradshaw, W. E., & Holzapfel, C. M. (2009). Complications of complexity: integrating environmental, genetic and hormonal control of insect diapause. *Trends in Genetics* 25, 217-225.
- Fairbairn, D. J. (2011). The advent of mandatory data archiving. *Evolution* 65, 1-2.
- Finley, C. M., Gorman, M. R., Tuthill, C. R., & Zucker, I. (1995). Long-term reproductive effects of a single long day in the siberian hamster (*Phodopus* sungorus). Journal of Biological Rhythms 10, 33-41.
- Gekakis, N., Staknis, D., Nguyen, H. B., Davis, F. C., Wilsbacher, L. D., King, D. P., Takahashi, J. S., & Weitz, C. J. (1998). Role of the CLOCK protein in the mammalian circadian mechanism. *Science* 280, 1564-1569.
- Goldman, B. D. (2001). Mammalian photoperiodic system: formal properties and neuroendocrine mechanisms of photoperiodic time measurement. *Journal of Biological Rhythms* **16**, 283-301.
- Goldman, B. D. B., Gwinner, E., Karsch, F. J., Saunders, D. S., Zucker, I., & Gall, G. F. (2004). Chronobiology: Biological Timekeeping. In: *Circannual rhythms and photoperiodism*, (Dunlap, J. C., Loros, J. J., & Decoursey, P. J., eds), pp. 107-142. Sunderland: Sinauer Associates.
- Goto, S. G., Han, B., & Denlinger, D. L. (2006). A nondiapausing variant of the flesh fly, Sarcophaga bullata, that shows arrhythmic adult eclosion and elevated expression of two circadian clock genes, *period* and *timeless*. *Journal of Insect Physiology* 52, 1213-1218.
- Goto, S. G., Shiga S. & Numata H. (2010). Photoperiodism in insects: perception of light and the role of clock genes. In *Photoperiodism: The Biological Calendar* (Nelson, R. J., Denlinger, D. L. & Somers, D. E., eds), pp. 258-286, Oxford: Oxford University Press.
- Guderley, H., Leroy, P. H., & Gagné, A. (2001). Thermal acclimation, growth, and burst swimming of threespine stickleback: enzymatic correlates and influence of photoperiod. *Physiological and Biochemical Zoology* 74, 66-74.

- Guindon, S., Delsuc, F., Dufayard, J., & Gascuel, O. (2009). Estimating maximum likelihood phylogenies with PhyML. *Methods in Molecular Biology* 537, 107-142.
- Halford, S., Pires, S. S., Turton, M., Zheng, L., García-Menéndez, I., Davies, W. L., Peirson, S. N., Garcla-Fern ndez, J. M., Hankins, M. W., & Foster, R. G. (2009). VA opsin-based photoreceptors in the hypothalamus of birds. *Current Biology* 19, 1396-1402.
- Han, B. & Denlinger, D. L. (2009). Length variation in a specific region of the period gene correlates with differences in pupal diapause incidence in the flesh fly, *Sarcophaga bullata. Journal of Insect Physiology* 55, 415-418.
- Hanon, E. A., Lincoln, G. A., Fustin, J. M., Dardente, H., Masson-Pévet, M., Morgan, P. J., & Hazlerigg, D. G. (2008). Ancestral TSH mechanism signals summer in a photoperiodic mammal. *Current Biology* 18, 1147-1152.
- Hazlerigg, D. & Loudon, A. (2008). New insights into ancient seasonal life timers. *Current Biology* 18, R795-R804.
- Heideman, P. D., Bruno, T. A., Singley, J. W., & Smedley, J. V. (1999). Genetic variation in photoperiodism in *Peromyscus leucopus*: geographic variation in an alternative life-history strategy. *Journal of Mammalogy* 80, 1232-1242.
- Helfer, G., Fidler, A. E., Vallone, D., Foulkes, N. S., & Brandstaetter, R. (2006). Molecular analysis of clock gene expression in the avian brain. *Chronobiology International* 23, 113-127.
- Hellqvist, A., Bornestaf, C., Borg, B., & Schmitz, M. (2004). Cloning and sequencing of the FSH-β and LH β-subunit in the three-spined stickleback, *Gasterosteus aculeatus*, and effects of photoperiod and temperature on LH-β and FSH-β mRNA expression. *General and Comparative Endocrinology* **135**, 167-174.
- Hellqvist, A., Schmitz, M., Mayer, I., & Borg, B. (2006). Seasonal changes in expression of LH-β and FSH-β in male and female three-spined stickleback, *Gasterosteus aculeatus*. *General and Comparative Endocrinology* **145**, 263-269.
- Hendry, A. P., Taylor, E. B., McPhail, D. J., & Karl, S. (2002). Adaptive divergence and the balance between selection and gene flow: lake and stream stickleback in the misty system. *Evolution* **56**, 1199-1216.
- Heyland, A., Hodin, J., & Reitzel, A. M. (2005). Hormone signaling in evolution and development: a non-model system approaches. *BioEssays* 27, 64-75.
- Hibbeler, S., Scharsack, J., & Becker, S. (2008). Housekeeping genes for quantitative expression studies in the three-spined stickleback *Gasterosteus aculeatus*. *BMC Molecular Biology* 9, 18.
- Hiney, J. K., Sower, S. A., Yu, W. H., McCann, S. M., & Dees, W. L. (2002). Gonadotropin-releasing hormone neurons in the preoptic-hypothalamic region of the rat contain lamprey gonadotropin-releasing hormone III, mammalian luteinizing hormone-releasing hormone, or both peptides. *Proceedings of the National Academy of Sciences USA* **99**, 2386-2391.

- Hohenlohe, P. A., Bassham, S., Etter, P. D., Stiffler, N., Johnson, E. A., & Cresko, W. A. (2010). Population genomics of parallel adaptation in threespine stickleback using sequenced RAD tags. *PLoS Genetics* 6, e1000862.
- Ikeno, T., Tanaka, S., Numata, H., & Goto, S. (2010). Photoperiodic diapause under the control of circadian clock genes in an insect. *BMC Biology* **8**, 116.
- Jin, X., Shearman, L. P., Weaver, D. R., Zylka, M. J., De Vries, G. J., & Reppert, S. M. (1999). A molecular mechanism regulating rhythmic output from the suprachiasmatic circadian clock. *Cell* 96, 57-68.
- Johnsen, A., Fidler, A. E., Kuhn, S., Carte, K. L., Hoffmann, A., Barr, I. R., Biard, C., Charmantier, A., Eens, M., Korsten, P., Siitari, H., Tomiuk, J., & Kempenaers, B. (2007). Avian Clock gene polymorphism: evidence for a latitudinal cline in allele frequencies. *Molecular Ecology* 16, 4867-4880.
- Jones, D. T., Taylor, W. R., & Thornton, J. M. (1992). The rapid generation of mutation data matrices from protein sequences. *Computer applications in the biosciences : CABIOS* **8**, 275 282.
- Kah, O. & Dufour, S. (2010). Conserved and divergent features of reproductive neuroendocrinology in teleost fishes. In: *Hormones and Reproduction of Vertebrates*, (Norris, D. O. & Lopez, K. H. eds), pp. 15-42. New York: Elsevier Press.
- Kah, O., Lethimonier, C., Somoza, G., Guilgur, L. G., Vaillant, C., & Lareyre, J. J. (2007). GnRH and GnRH receptors in metazoa: A historical, comparative, and evolutive perspective. *General and Comparative Endocrinology* 153, 346-364.
- Kanda, S., Okubo, K., & Oka, Y. (2011). Differential regulation of the luteinizing hormone genes in teleosts and tetrapods due to their distinct genomic environments - Insights into gonadotropin beta subunit evolution. *General and Comparative Endocrinology* **173**, 253 - 258.
- Karvé, A., von Hippel, F., & Bell, M. (2008). Isolation between sympatric anadromous and resident threespine stickleback species in Mud Lake, Alaska. *Environmental Biology of Fishes* 81, 287 - 296.
- Kasper, R., Shved, N., Takahashi, A., Reinecke, M., & Eppler, E. (2006). A systematic immunohistochemical survey of the distribution patterns of GH, prolactin, somatolactin, β--TSH, β--FSH, β--LH, ACTH, and α--MSH in the adenohypophysis of *Oreochromis niloticus*, the Nile tilapia. *Cell and Tissue Research* **325**, 303-313.
- Kemp, A. (1984). Spawning of the Australian lungfish, *Neoceratodus fosteri* (Krefft) in the Brisbane River and Enoggera Reservoir, Queensland. *Memoirs of the Queensland Museum* 21, 391-399.
- King, D. P., Zhao, Y., Sangoram, A. M., Wilsbacher, L. D., Tanaka, M., Antoch, M. P., Steeves, T. D. L., Vitaterna, M. H., Kornhauser, J. M., Lowrey, P. L., Turek, F. W., & Takahashi, J. S. (1997). Positional cloning of the mouse circadian *Clock* gene. *Cell* 89, 641-653.

- Kingsley, D. M., Zhu, B., Osoegawa, K., De Jong, P. J., Schein, J., Marra, M., Peichel, C., Amemiya, C., Schluter, D., Balabhadra, S., Friedlander, B., Yee Man, C., Dickson, M., Grimwood, J., Schmutz, J., Talbot, W. S., & Myers, R. (2004). New genomic tools for molecular studies of evolutionary change in threespine sticklebacks. *Behaviour* 141, 1331-1344.
- Kingsolver, J. G. & Schemske, D. W. (1991). Path analyses of selection. *Trends in Ecology & Evolution* 6, 276-280.
- Kitano, J., Lema, S. C., Luckenbach, J. A., Mori, S., Kawagishi, Y., Kusakabe, M., Swanson, P., & Peichel, C. L. (2010). Adaptive divergence in the thyroid hormone signaling pathway in the stickleback radiation. *Current Biology* 20, 2124-2130.
- Kobayashi, Y. & Weigel, D. (2007). Move on up, it's time for change---mobile signals controlling photoperiod-dependent flowering. *Genes & Development* **21**, 2371-2384.
- Koštál, V. (2011). Insect photoperiodic calendar and circadian clock: independence, cooperation, or unity? *Journal of Insect Physiology* **57**, 538 556.
- Kuo, C. M., Nash, C. E., & Shehadeh, Z. H. (1974). The effects of temperature and photoperiod on ovarian development in captive grey mullet (*Mugil cephalus* L). *Aquaculture* 3, 25-43.
- Kyriacou, C. P., Peixoto, A. A., Sandrelli, F., Costa, R., & Tauber, E. (2008). Clines in clock genes: fine-tuning circadian rhythms to the environment. *Trends in Genetics* 24, 124-132.
- Labbé, P., Sidos, N., Raymond, M., & Lenormand, T. (May 2009). Resistance gene replacement in the mosquito *Culex pipiens*: fitness estimation from long-term cline series. *Genetics* **182**, 303-312.
- Lambrechts, M. M., Blondel, J., Maistre, M., & Perret, P. (1997). A single response mechanism is responsible for evolutionary adaptive variation in a bird's laying date. *Proceedings of the National Academy of Sciences USA* 94, 5153-5155.
- Lankinen, P. (1986). Geographical variation in circadian eclosion rhythm and photoperiodic adult diapause *Drosophila littoralis*. *Journal of Comparative Physiology A: Neuroethology, Sensory, Neural, and Behavioral Physiology* 159, 123-142.
- Lankinen, P. & Forsman, P. (2006). Independence of genetic geographical variation between photoperiodic diapause, circadian eclosion rhythm, and Thr-Gly repeat region of the period gene in *Drosophila littoralis*. *Journal of Biological Rhythms* 21, 3-12.
- Lashbrook, M. K. & Livezey, R. L. (1970). Effects of photoperiod on heat tolerance in *Sceloporus occidentalis occidentalis. Physiological Zoology* **43**, 38-46.

- Laurila, A., Pakkasmaa, S., & Merilä, J. (2001). Influence of seasonal time constraints on growth and development of common frog tadpoles: a photoperiod experiment. *Oikos* 95, 451-460.
- Leska, A. & Dusza, L. (2007). Seasonal changes in the hypothalamo-pituitarygonadal axis in birds. *Reproductive Biology* **7**, 99 126.
- Lewontin, R. C. (1974). Genetic Basis of Evolutionary Change. New York: Columbia University Press.
- Li, W-S. (1997). Molecular Evolution. Amherst, MA: Sinauer Associates.
- Liedvogel, M. & Sheldon, B. C. (2010). Low variability and absence of phenotypic correlates of Clock gene variation in a great tit Parus major population. *Journal of Avian Biology* **41**, 543--550.
- Liedvogel, M., Szulkin, M., Knowles, S. C. L., Wood, M. J., & Sheldon, B. C. (2009). Phenotypic correlates of *Clock* gene variation in a wild blue tit population: evidence for a role in seasonal timing of reproduction. *Molecular Ecology* **18**, 2444-2456.
- Lowrey, P. L., Shimomura, K., Antoch, M. P., Yamazaki, S., Zemenides, P. D., Ralph, M. R., Menaker, M., & Takahashi, J. S. (2000). Positional syntenic cloning and functional characterization of the mammalian circadian mutation *tau*. *Science* 288, 483-491.
- Lowrey, P. L. & Takahashi, J. S. (2004). Mammalian circadian biology: elucidating genome-wide levels of temporal organization. *Annual Review of Genomics and Human Genetics* **5**, 407-441.
- Lynch, G. R., Heath, H. W., & Johnston, C. M. (1981). Effect of geographical origin on the photoperiodic control of reproduction in the white-footed Mouse, *Peromyscus leucopus*. *Biology of Reproduction* 25, 475-480.
- Majoy, S. B. & Heideman, P. D. (2000). *Tau* differences between short-day responsive and short-day nonresponsive white-footed mice (*Peromyscus leucopus*) do not affect reproductive photoresponsiveness. *Journal of Biological Rhythms* 15, 500-512.
- Masuda, T., Iigo, M., & Aida, K. (2005). Existence of an extra-retinal and extra-pineal photoreceptive organ that regulates photoperiodism in gonadal development of an Osmerid teleost, ayu (*Plecoglossus altivelis*). Comparative Biochemistry and Physiology - Part A: Molecular & Integrative Physiology 140, 414-422.
- Masumoto, K. H., Ukai-Tadenuma, M., Kasukawa, T., Nagano, M., Uno, K. D., Tsujino, K., Horikawa, K., Shigeyoshi, Y., & Ueda, H. R. (2010). Acute induction of *eya3* by late-night light stimulation triggers TSHβ expression in photoperiodism. *Current Biology* **20**, 2199-2206.
- Mathias, D., Jacky, L., Bradshaw, W. E., & Holzapfel, C. M. (2007). Quantitative trait loci associated with photoperiodic response and stage of diapause in the pitcherplant mosquito. *Genetics* 176, 391-402.

- Mayer, I., Borg, B., & Pall, M. (2004). Hormonal control of male reproductive behavior in fishes: a stickleback perspective. *Behaviour* **141**, 1151-1162.
- Mayer, I., Bornestaf, C., & Borg, B. (1997). Melatonin in non-mammalian vertebrates: Physiological role in reproduction? *Comparative Biochemistry and Physiology Part A: Physiology* **118**, 515-531.
- McInernev, J. E. & Evans, D. O. (1970). Action spectrum of the photoperiod mechanism controlling sexual maturation in the threespine stickleback, *Gasterosteus aculeatus*. Journal of the Fisheries Research Board of Canada **27**, 749-763.
- McKechnie, S. W., Blacket, M. J., Song, S. V., Rako, L., Carroll, X., Johnson, T. K., Jensen, L. T., Lee, S. F., Wee, C. W., & Hoffmann, A. A. (2010). A clinally varying promoter polymorphism associated with adaptive variation in wing size in *Drosophila*. *Molecular Ecology* 19, 775-784.
- Michael, T. P., Park, S., Kim, T.-S., Booth, J., Byer, A., Sun, Q., Chory, J., & Lee, K. (2007). Simple sequence repeats provide a substrate for phenotypic variation in the *Neurospora crassa* circadian clock. *PLoS ONE* 2, e795.
- Migaud, H., Davie, A., & Taylor, J. F. (2010). Current knowledge on the photoneuroendocrine regulation of reproduction in temperate fish species. *Journal dasof Fish Biology* **76**, 27--68.
- Millar, R. P. (2005). GnRHs and GnRH receptors. *Animal Reproduction Science* **88**, 5-28.
- Miranda, L., Strüssmann, C., & Somoza, G. (2009). Effects of light and temperature conditions on the expression of GnRH and GtH genes and levels of plasma steroids in *Odontesthes bonariensis* females. *Fish Physiology and Biochemistry* 35, 101-108.
- Moore, R. Y. (1995). Neural control of the pineal gland. *Behavioural Brain Research* **73**, 125-130.
- Møller, A. P. (2007). Tardy females, impatient males: protandry and divergent selection on arrival date in the two sexes of the barn swallow. *Behavioral Ecology and Sociobiology* **61**, 1311-1319.
- Nakane, Y., Ikegami, K., Ono, H., Yamamoto, N., Yoshida, S., Hirunagi, K., Ebihara, S., Kubo, Y., & Yoshimura, T. (2010). A mammalian neural tissue opsin (Opsin 5) is a deep brain photoreceptor in birds. *Proceedings of the National Academy of Sciences USA* 107, 15264-15268.
- Nakao, N., Ono, H., Yamamura, T., Anraku, T., Takagi, T., Higashi, K., Yasuo, S., Katou, Y., Kageyama, S., Uno, Y., Kasukawa, T., Iigo, M., Sharp, P. J., Iwasawa, A., Suzuki, Y., Sugano, S., Niimi, T., Mizutani, M., Namikawa, T., Ebihara, S., Ueda, H. R., & Yoshimura, T. (2008). Thyrotrophin in the pars tuberalis triggers photoperiodic response. *Nature* 452, 317-322.
- Nocedal, J. (1980). Updating quasi-newton matrices with limited storage. *Mathematics of Computation* **35**, 773-782.

- Nocillado, J. N., Levavi-Sivan, B., Carrick, F., & Elizur, A. (2007). Temporal expression of G-protein-coupled receptor 54 (GPR54), gonadotropin-releasing hormones (GnRH), and dopamine receptor D2 (drd2) in pubertal female grey mullet, *Mugil cephalus. General and Comparative Endocrinology* **150**, 278 - 287.
- Norberg, B., Brown, C. L., Halldorsson, O., Stensland, K., & Björnsson, B. T. (2004). Photoperiod regulates the timing of sexual maturation, spawning, sex steroid and thyroid hormone profiles in the Atlantic cod (*Gadus morhua*). Aquaculture 229, 451 - 467.
- O'Brien, C. S., Bradshaw, W. E., & Holzapfel, C. M. (2011). Testing for causality in covarying traits: genes and latitude in a molecular world. *Molecular Ecology* **20**, 2471--2476.
- O'Brien, C. S., Unruh, L. A., Zimmerman, C. E., Bradshaw, W. E., Holzapfel, C. M., & Cresko, W. A. (). Geography of the circadian gene *clock* and photoperiodic response in western North American populations of the threespine stickleback, *Gasterosteus aculeatus*. *In prep*.
- O'Malley, K. G. & Banks, M. A. (2008a). A latitudinal cline in the Chinook salmon (*Oncorhynchus tshawytscha*) *Clock* gene: evidence for selection on polyQ length variants. *Proceedings of the Royal Society B: Biological Sciences* **275**, 2813-2821.
- O'Malley, K. & Banks, M. (2008b). Duplicated *Clock* genes with unique polyglutamine domains provide evidence for nonhomologous recombination in Chinook salmon (*Oncorhynchus tshawytscha*). *Genetica* 132, 87-94.
- O'Malley, K. G., Ford, M. J., & Hard, J. J. (2010). *Clock* polymorphism in Pacific salmon: evidence for variable selection along a latitudinal gradient. *Proceedings* of the Royal Society B: Biological Sciences **277**, 3703-3714.
- Okubo, K. & Nagahama, Y. (2008). Structural and functional evolution of gonadotropinreleasing hormone in vertebrates. *Acta Physiologica* **193**, 3-15.
- Ono, H., Hoshino, Y., Yasuo, S., Watanabe, M., Nakane, Y., Murai, A., Ebihara, S., Korf, H. W., & Yoshimura, T. (2008). Involvement of thyrotropin in photoperiodic signal transduction in mice. *Proceedings of the National Academy* of Sciences USA 105, 18238-18242.
- Orozco, A. & Valverde R. C. (2005). Thyroid hormone deiodination in fish. *Thyroid* **15**, 799-813.
- Paaby, A. B., Blacket, M. J., Hoffmann, A. A., & Schmidt, P. S. (2010). Identification of a candidate adaptive polymorphism for *Drosophila* life history by parallel independent clines on two continents. *Molecular Ecology* 19, 760-774.
- Parhar, I. S., Soga, T., & Sakuma, Y. (2000). Thyroid hormone and estrogen regulate brain region-specific messenger ribonucleic acids encoding three gonadotropinreleasing hormone genes in sexually immature male fish, *Oreochromis niloticus*. *Endocrinology* 141, 1618-1626.

- Paris, M., Hillenweck, A., Bertrand, S., Delous, G., Escriva, H., Zalko, D., Cravedi, J. P., & Laudet, V. (2010). Active metabolism of thyroid hormone during metamorphosis of amphioxus. *Integrative and Comparative Biology* 50, 63-74.
- Petraitis, P. S., Dunham, A. E., & Niewiarowski, P. H. (1996). Inferring multiple causality: the limitations of path analysis. *Functional Ecology* **10**, 421-431.
- Philp, A. R., Garcia-Fernandez, J. M., Soni, B. G., Lucas, R. J., Bellingham, J., & Foster, R. G. (2000). Vertebrate ancient (VA) opsin and extraretinal photoreception in the Atlantic salmon (Salmo salar). *Journal of Experimental Biology* 203, 1925 - 1936.
- Postlethwait, J., Amores, A., Cresko, W., Singer, A., & Yan, Y. L. (2004). Subfunction partitioning, the teleost radiation and the annotation of the human genome. *Trends in Genetics* **20**, 481-490.
- Proudman, J. A., Scanes, C. G., Johannsen, S. A., Berghman, L. R., & Camp, M. J. (2006). Comparison of the ability of the three endogenous GnRHs to stimulate release of follicle-stimulating hormone and luteinizing hormone in chickens. *Domestic Animal Endocrinology* **31**, 141-153.
- Proudman, J. A., Vandesande, F., & Berghman, L. R. (1999). Immunohistochemical evidence that follicle-stimulating hormone and luteinizing hormone reside in separate cells in the chicken pituitary. *Biology of Reproduction* 60, 1324-1328.
- Puebla-Osorio, N., Proudman, J. A., Compton, A. E., Clements, K. E., Decuypere, E., Vandesande, F., & Berghman, L. R. (2002). FSH- and LH-cells originate as separate cell populations and at different embryonic stages in the chicken embryo. *General and Comparative Endocrinology* 127, 242-248.
- Quinn, T. P. & Adams, D. J. (1996). Environmental changes affecting the migratory timing of American shad and Sockeye salmon. *Ecology* **77**, 1151-1162.
- Quinn, T. P., Unwin, M. J., & Kinnison, M. T. (2000). Evolution of temporal isolation in the wild: genetic divergence in timing of migration and breeding introduced Chinook salmon populations. *Evolution* 54, 1372-1385.
- Raine, J. C. (2010). Thyroid hormones and reproduction in fishes. In: *Hormones and Reproduction of Vertebrates*, (Norris, D. O. & Lopez, K. H. eds), pp. 83-102. New York: Elsevier Press.
- Rand, D. M., Weinreich, D. M., Lerman, D., Folk, D., & Gilchrist, G. W. (2010). Three selections are better than one: clinal variation of thermal QTL from independent selection experiments in *Drosophila*. *Evolution* 64, 2921-2934.
- Rosenblum, E. B., Römpler, H., Schöneberg, T., & Hoekstra, H. E. (2010). Molecular and functional basis of phenotypic convergence in white lizards at White Sands. *Proceedings of the National Academy of Sciences USA* **107**, 2113-2117.
- Réale, D., McAdam, A. G., Boutin, S., & Berteaux, D. (2003). Genetic and plastic responses of a northern mammal to climate change. *Proceedings of the Royal Society of London. Series B: Biological Sciences* 270, 591-596.
- Sall, J., Creighton, L. & Lehman, A. (2005). *JMP Start Statistics*, 3rd edn. Cary, NC: SAS Institute Inc.

- Saunders, D. S. (1990). The circadian basis of ovarian diapause regulation in *Drosophila melanogaster*: Is the period gene causally involved in photoperiodic time measurement?. *Journal of Biological Rhythms* 5, 315-331.
- Saunders, D. S. (2002). *Insect Clocks*, 3rd edn. Amsterdam, Netherlands: Elsevier Science.
- Saunders, D. (2010). Controversial aspects of photoperiodism in insects and mites. *Journal of Insect Physiology* **56**, 1491-1502.
- Saunders, D. S. & Bertossa, R. C. (2011). Deciphering time measurement: The role of circadian 'clock' genes and formal experimentation in insect photoperiodism. *Journal of Insect Physiology* 57, 557-566.
- Schiesari, L., Kyriacou, C. P., & Costa, R. (2011). The hormonal and circadian basis for insect photoperiodic timing. *FEBS Letters* 585, 1450 - 1460.
- Schluter, D., Clifford, E. A., Nemethy, M., & McKinnon, J. S. (2004). Parallel evolution and inheritance of quantitative traits.. *The American Naturalist* **163**, 809-822.
- Schluter, D., Marchinko, K. B., Barrett, R. D. H., & Rogers, S. M. (2010). Natural selection and the genetics of adaptation in threespine stickleback. *Philosophical Transactions of the Royal Society B: Biological Sciences* 365, 2479-2486.
- Schmidt, P. S., Zhu, C. T., Das, J., Batavia, M., Yang, L., & Eanes, W. F. (2008). An amino acid polymorphism in the couch potato gene forms the basis for climatic adaptation in Drosophila melanogaster. Proceedings of the National Academy of Sciences USA 105, 16207-16211.
- Sharp, P. J. (2005). Photoperiodic regulation of seasonal breeding in birds. *Annals of the New York Academy of Sciences* **1040**, 189-199.
- Sharp, P. J., Dawson, A., & Lea, R. W. (1998). Control of luteinizing hormone and prolactin secretion in birds. *Comparative Biochemistry and Physiology Part C: Pharmacology, Toxicology and Endocrinology* **119**, 275-282.
- Shimomura, K., Nelson, D., Ihara, N., & Menaker, M. (1997). Photoperiodic time measurement in *tau* mutant hamsters. *Journal of Biological Rhythms* 12, 423-430.
- Shupnik, M. A. (1996). Gonadal hormone feedback on pituitary gonadotropin genes. *Trends in Endocrinology and Metabolism* 7, 272-276.
- Silverin, B., Kikuchi, M., & Ishii, S. (1999). Effect of season and photoperiod on FSH in male great tits. *General and Comparative Endocrinology* **113**, 457-463.
- Sokal, R. R. & Rohlf, F. J. (1995). *Biometry: The Principles and Practice of Statistics in Biological Research*. New York: Freeman.
- Stehlík, J., Závodská, R., Shimada, K., Šauman, I., & Koštál, V. (2008). Photoperiodic induction of diapause requires regulated transcription of timeless in the larval brain of *Chymomyza costata*. Journal of Biological Rhythms 23, 129-139.

- Stinchcombe, J. R., Weinig, C., Ungerer, M., Olsen, K. M., Mays, C., Halldorsdottir, S. S., Purugganan, M. D., & Schmitt, J. (2004). A latitudinal cline in flowering time in *Arabidopsis thaliana* modulated by the flowering time gene FRIGIDA. *Proceedings of the National Academy of Sciences USA* 101, 4712-4717.
- Storz, J. F. & Wheat, C. W. (2010). Integrating evolutionary and functional approaches to infer adaptation at specific loci. *Evolution* 64, 2489-509.
- Sullivan, J. K. & Lynch, G. R. (1986). Photoperiod time measurement for activity, torpor, molt and reproduction in mice. *Physiology & Behavior* **36**, 167-174.
- Sánchez-Gonzáles, S., Ruiz-Campos, G., & Contreras-Balderas, S. (2001). Feeding ecology and habitat of the threespine stickleback, *Gasterosteus aculeatus microcephalus*, in a remnant population of northwestern Baja California, México. *Ecology of Freshwater Fish* 10, 191-197.
- Tauber, E. & Kyriacou, B. P. (2001). Insect photoperiodism and circadian clocks: models and mechanisms. *Journal of Biological Rhythms* 16, 381-390.
- Tauber, E., Zordan, M., Sandrelli, F., Pegoraro, M., Osterwalder, N., Breda, C., Daga, A., Selmin, A., Monger, K., Benna, C., Rosato, E., Kyriacou, C. P., & Costa, R. (2007). Natural selection favors a newly derived *timeless* allele in *Drosophila melanogaster*. *Science* **316**, 1895-1898.
- R Development Core Team (2007). *A Language and Environment for Statistical Computing*. Vienna: R Foundation for Statistical Computing.
- Thisse, C., Thisse, B., Schilling, T. F., & Postlethwait, J. H. (1993). Structure of the zebrafish snail1 gene and its expression in wild-type, spadetail and no tail mutant embryos. *Development* **119**, 1203-1215.
- Thom, M., Johnson, D., & MacDonald, D. (2004). The evolution and maintenance of delayed implantation in the mustelidae (Mammalia : Carnivora). *Evolution* **58**, 175-183.
- Uller, T. & Olsson, M. (2003). Life in the land of the midnight sun: are northern lizards adapted to longer days? *Oikos* 101, 317-322.
- U. S. Department of Commerce (1968). *Climatic Atlas of the United States*. Washington: Environmental Science Services Administration, Environmental Data Service, U. S. Government Printing Office.
- Vatine, G., Vallone, D., Gothilf, Y., & Foulkes, N. S. (2011). It's time to swim! Zebrafish and the circadian clock. *FEBS Letters* **585**, 1485 1494.
- Vaz Nunes, M. & Saunders, D. (1999). Photoperiodic time measurement in insects: a review of clock models. *Journal of Biological Rhythms* 14, 84-104.
- Watanabe, M., Yasuo, S., Watanabe, T., Yamamura, T., Nakao, N., Ebihara, S., & Yoshimura, T. (2004). Photoperiodic regulation of type 2 deiodinase gene in Djungarian hamster: possible homologies between avian and mammalian photoperiodic regulation of reproduction. *Endocrinology* 145, 1546-1549.

- Weaver, D. R., Liu, C., & Reppert, S. M. (1996). Nature's knockout: the mel1b receptor is not necessary for reproductive and circadian responses to melatonin in Siberian hamsters. *Molecular Endocrinology* **10**, 1478-1487.
- Westwood, R. K. & Murton, N. J. (1977). *Avian Breeding Cycles*. London: Oxford University Press.
- Wilczek, A. M., Roe, J. L., Knapp, M. C., Cooper, M. D., Lopez-Gallego, C., Martin, L. J., Muir, C. D., Sim, S., Walker, A., Anderson, J., Egan, J. F., Moyers, B. T., Petipas, R., Giakountis, A., Charbit, E., Coupland, G., Welch, S. M., & Schmitt, J. (2009). Effects of genetic perturbation on seasonal life history plasticity. *Science* 323, 930-934.
- Yahalom, D., Chen, A., Ben-Aroya, N., Rahimipour, S., Kaganovsky, E., Okon, E., Fridkin, M., & Koch, Y. (1999). The gonadotropin-releasing hormone family of neuropeptides in the brain of human, bovine and rat: identification of a third isoform. *FEBS Letters* 463, 289-294.
- Yamamura, T., Hirunagi, K., Ebihara, S., & Yoshimura, T. (2004). Seasonal morphological changes in the neuro-glial interaction between gonadotropinreleasing hormone nerve terminals and glial endfeet in Japanese quail. *Endocrinology* 145, 4264-4267.
- Yamamura, T., Yasuo, S., Hirunagi, K., Ebihara, S., & Yoshimura, T. (2006). T3 implantation mimics photoperiodically reduced encasement of nerve terminals by glial processes in the median eminence of Japanese quail. *Cell and Tissue Research* 324, 175-179.
- Yang, Z. (1996). Among-site rate variation and its impact on phylogenetic analyses. *Trends in Ecology & Evolution* **11**, 367-372.
- Yasuo, S., Watanabe, M., Iigo, M., Nakamura, T. J., Watanabe, T., Takagi, T., Ono, H., Ebihara, S., & Yoshimura, T. (2007). Differential response of type 2 deiodinase gene expression to photoperiod between photoperiodic Fischer 344 and nonphotoperiodic Wistar rats. *American Journal of Physiology - Regulatory*, *Integrative and Comparative Physiology* 292, R1315-R1319.
- Yasuo, S., Watanabe, M., Nakao, N., Takagi, T., Follett, B. K., Ebihara, S., & Yoshimura, T. (2005). The reciprocal switching of two thyroid hormoneactivating and -inactivating enzyme genes is involved in the photoperiodic gonadal response of Japanese quail. *Endocrinology* 146, 2551-2554.
- Yasuo, S. & Yoshimura, T. (2009). Comparative analysis of the molecular basis of photoperiodic signal transduction in vertebrates. *Integrative and Comparative Biology* 49, 507-518.
- Yasuo, S., Yoshimura, T., Ebihara, S., & Korf, H.-W. (2010). Photoperiodic control of TSH-β expression in the mammalian pars tuberalis has different impacts on the induction and suppression of the hypothalamo-hypopysial gonadal axis. *Journal of Neuroendocrinology* **22**, 43-50.

- Yeates-Burghart, Q. S., O'Brien, C., Cresko, W. A., Holzapfel, C. M., & Bradshaw, W. E. (2009). Latitudinal variation in photoperiodic response of the three-spined stickleback *Gasterosteus aculeatus* in western North America. *Journal of Fish Biology* **75**, 2075-2081.
- Yerushalmi, S. & Green, R. M. (2009). Evidence for the adaptive significance of circadian rhythms. *Ecology Letters* 12, 970-981.
- Yerushalmi, S., Yakir, E., & Green, R. M. (2011). Circadian clocks and adaptation in Arabidopsis. Molecular Ecology 20, 1155 - 1165.
- Yoshimura, T. (2010). Neuroendocrine mechanism of seasonal reproduction in birds and mammals. *Animal Science Journal* **8**, 403-410.
- Yoshimura, T., Yasuo, S., Watanabe, M., Iigo, M., Yamamura, T., Hirunagi, K., & Ebihara, S. (2003). Light-induced hormone conversion of T4 to T3 regulates photoperiodic response of gonads in birds. *Nature* **426**, 178-181.
- Zar, J. H. (1996). Biostatistical Analysis. New Jersey: Prentice Hall.
- Zhang, Q., Li, H., Li, R., Hu, R., Fan, C., Chen, F., Wang, Z., Liu, X., Fu, Y., & Lin, C. (2008). Association of the circadian rhythmic expression of GmCRY1a with a latitudinal cline in photoperiodic flowering of soybean. *Proceedings of the National Academy of Sciences USA* 105, 21028-21033.